# SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM S-3
REGISTRATION STATEMENT
UNDER
THE SECURITIES ACT OF 1933

### **3M COMPANY**

(Exact name of Registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 41-0417775 (I.R.S. Employer Identification No.)

3M Center St. Paul, Minnesota 55144 (651) 733-2204

(Address, including zip code, and telephone number, including area code, of Registrant's principal executive offices)

Gregg M. Larson Assistant General Counsel and Secretary 3M Company 3M Center St. Paul, Minnesota 55144 (651) 733-2204

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Approximate date of commencement of proposed sale to the public: From time to time after the effective date of this registration statement.

If the only securities being registered on this form are being offered pursuant to dividend or interest reinvestment plans, please check the following box. 

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, other than securities offered only in connection with dividend or interest reinvestment plans, check the following box. 

If this form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this form is a post-effective amendment filed pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If delivery of the prospectus is expected to be made pursuant to Rule 434, please check the following box.

Title of Each Class of Securities to be Registered	Number of shares To be Registered	Proposed Maximum Offering Price Per Share (1)	Proposed Maximum Aggregate Offering Price (1)	Amount of Registration Fee	
Common Stock, par value of \$0.01 per share	541,581	\$78.585	\$42,560,143	\$5,393	

<sup>(1)</sup> Estimated in accordance with Rule 457(c) solely for the purposes of calculating the registration fee, based on the average of the high and low sales prices for the common stock reported on the New York Stock Exchange on February 2, 2004.

The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this registration statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the registration statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

The information in this prospectus is not complete and may be changed. The selling security holders may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any state where the offer or sale is not permitted.

SUBJECT TO COMPLETION, DATED FEBRUARY 6, 2004

### **3M COMPANY**

### 541,581 SHARES Common Stock

This prospectus relates to the public offering, which is not being underwritten, of 541,581 shares of our common stock, par value \$0.01 per share. Selling stockholders named in this document together with any of their pledgees, donees, transferees or other successors in interest that receive such shares as a gift, partnership distribution or other non-sale related transfer are collectively known as the selling stockholders. Selling stockholders may offer the shares from time to time.

The prices at which such selling stockholders may sell the shares will be determined by the prevailing market price for the shares or in negotiated transactions. We will not receive any of the proceeds from the sale of the shares.

We originally issued the shares in connection with an Amended Agreement and Plan of Merger among Steeler Meger Corporation, Steeler Meger LLC, HighJump Software, Inc. and us, dated January 14, 2004. We issued the shares pursuant to an exemption from the registration requirements of the Securities Act of 1933, as amended (the "Securities Act"), provided under its Section 4(2). We are registering the shares pursuant to the Amended Agreement and Plan of Merger.

Our common stock is listed on the New York, Chicago, Pacific and Swiss stock exchanges under the symbol "MMM." On February 2, 2004, the average of the high and low price for the common stock on the New York Stock Exchange was \$78.585.

See "Risk Factors" beginning on page 3 for factors you should consider before buying shares of our common stock.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or passed	upon the
accuracy or adequacy of this prospectus. Any representation to the contrary is a criminal offense.	

04
,

You should rely only on the information contained or incorporated by reference in this prospectus and in any accompanying prospectus supplement. No one has been authorized to provide you with different information.

You should not assume that the information in this prospectus or any prospectus supplement is accurate as of any date other that the date on the front of the documents

The shares of common stock are not being offered in any jurisdiction where the offer is not permitted.

### THE COMPANY

3M was incorporated in 1929 under the laws of the State of Delaware to continue operations, begun in 1902, of a Minnesota corporation originally named Minnesota Mining and Manufacturing Company. On April 8, 2002, the company's name was changed to 3M Company. 3M's principal executive offices are located at 3M Center, St. Paul, Minnesota 55144 (telephone: 651-733-1110).

3M is a diversified global company with a market presence in health care, safety, electronics, telecommunications, industrial, consumer and office, and other markets. 3M is an integrated enterprise characterized by substantial intercompany cooperation in research, manufacturing and marketing of products. 3M's business has developed from its research and technology in coating and bonding for coated abrasives, the company's original product. Coating and bonding is the process of applying one material to another, such as abrasive granules to paper or cloth (coated abrasives), adhesives to a backing (pressure-sensitive tapes), ceramic coating to granular mineral (roofing granules), glass beads to plastic backing (reflective sheeting), and low-tack adhesives to paper (repositionable notes).

3M is among the leading manufacturers of products for many of the markets it serves. In all cases, 3M products are subject to direct or indirect competition. Most 3M products involve expertise in product development, manufacturing and marketing, and are subject to competition from products manufactured and sold by other technically oriented companies.

Our strategic business units have been aggregated into seven reportable segments: Industrial Business, Health Care Business, Transportation Business, Display and Graphics Business, Safety, Security and Protection Services Business, Consumer and Office Business, and Electro and Communications Business. These segments bring together common or related 3M technologies, enhancing the development of innovative products and services and providing for efficient sharing of business resources. These segments have worldwide responsibility for virtually all 3M product lines. A few miscellaneous businesses and staff-sponsored products, as well as various corporate assets and corporate overhead expenses, are not assigned to the segments.

#### FORWARD LOOKING INFORMATION

All statements included or incorporated by reference in this document, other than statements of historical facts, that address activities, events or developments that we intend, expect, project, believe or anticipate will or may occur in the future are forward looking statements. Such statements are typically characterized by terminology such as "believe," "anticipate," "should," "intend," "plan," "will," "expect," "estimate," "project," "positioned," "strategy," and similar expressions. These statements are based on assumptions and assessments made by our management in light of its experience and its perception of historical trends, current conditions, expected future developments and other factors our management believes to be appropriate. These forward looking statements are subject to a number of risks and uncertainties, including those risks described in this prospectus under "Risk Factors," as well as other factors that our management has not yet identified. Any such forward looking statements are not guarantees of future performance and actual results, developments and business decisions may differ from those contemplated by such forward looking statements. We disclaim any duty to update any forward looking statements.

### **RISK FACTORS**

An investment in our common stock involves a number of risks, some of which could be substantial and are inherent in our businesses. You should consider the following factors carefully before deciding to purchase shares of our common stock. Additional risks not presently known to 3M or that 3M currently deems immaterial may also impair 3M's business operations. Actual future results and trends may differ materially from historical results or those anticipated depending on a variety of factors, including, but not limited to, the following:

3M's business may be affected by the effects of, and changes in, worldwide economic conditions. 3M operates in more than 60 countries and derives more than half of its revenues from sales outside the United States. 3M's business may be affected by factors in the United States and other countries that are beyond its control, such as downturns in economic activity in a specific country or region; social, political or labor conditions in a specific country or region; or potential adverse foreign tax consequences.

Foreign currency exchange rates and fluctuations in those rates may affect 3M's ability to realize projected growth rates in its sales and net earnings and its results of operations. Because 3M derives more than half of its revenues from sales outside the United States, 3M's ability to realize projected growth rates in its sales and net earnings and its results of operations could be adversely affected if the United States dollar strengthens significantly against foreign currencies.

3M's growth objectives are largely dependent on the timing and market acceptance of its new product offerings, including its ability to renew its pipeline of new products and to bring those products to market. This ability may be adversely affected by difficulties or delays in product development, such as the inability to:

- identify viable new products;
- obtain adequate intellectual property protection;

3

- gain market acceptance of new products; or
- Successfully complete clinical trials and obtain regulatory approvals. For example, new 3M pharmaceutical products, like any pharmaceutical under
  development, face substantial risks and uncertainties in the process of development and regulatory review. There are no guarantees that products will receive
  regulatory approvals or prove to be commercially successful.

3M's future results are subject to fluctuations in the costs of purchased components and materials due to market demand, currency exchange risks, shortages and other factors. 3M depends on various components and materials for the manufacturing of its products. Although 3M has not experienced any difficulty in obtaining components and materials, it is possible that any of its supplier relationships could be terminated in the future. Any sustained interruption in 3M's receipt of adequate supplies could have a material adverse effect on it. In addition, while 3M has a process to minimize volatility in component and material pricing, no assurance can be given that 3M will be able to successfully manage price fluctuations due to market demand, currency risks, or shortages or that future price fluctuations will not have a material adverse effect on it.

The possibility that 3M's acquisitions, divestitures and strategic alliances may not meet sales and/or profit expectations. As part of 3M's strategy for growth, it has made and may continue to make acquisitions and divestitures and enter into strategic alliances. However, there can be no assurance that any of these will be completed or beneficial to 3M.

3M's future results may be affected by various legal and regulatory proceedings, including those involving product liability, antitrust, environmental or other subjects. The outcome of these legal proceedings may differ from 3M's expectations because the outcomes of litigation, including regulatory matters, are often difficult to predict reliably. Various factors or developments can lead 3M to change current estimates of liabilities and related insurance receivables where applicable, such as a final adverse judgment, significant settlement or changes in applicable law. A future adverse ruling or unfavorable development could result in future charges that could have a material adverse effect on 3M's consolidated financial position, results of operations or cash flows in any particular period. A specific factor that may influence 3M's estimate of its future asbestos-related liabilities is the currently pending Congressional consideration of legislation to reform asbestos-related litigation and pertinent information derived from that process.

For a more detailed discussion of legal proceedings involving 3M, see the discussion of "Legal Proceedings" in Part II, Item 1 of 3M's Quarterly Report on Form 10-Q for the period ended September 30, 2003, which is incorporated by reference into this prospectus.

3M files annual, quarterly and current reports, proxy statements and other information with the SEC. 3M's SEC filings are available to the public over the Internet at the SEC's web site at http://www.sec.gov. You may also read and copy any document 3M files with the SEC at its public reference room located at 450 Fifth Street, N.W., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the operation of the public reference facilities. 3M's SEC filings are also available at the office of the New York Stock Exchange. For further information on obtaining copies of 3M's public filings at the New York Stock Exchange, you should call (212) 656-5060

"Incorporated by reference" into this prospectus is the information in documents that 3M files with the SEC. This means that important information about 3M and 3M common stock is being disclosed to you by referring you to those documents. The information incorporated by reference is an important part of this prospectus, and you should rely on the information contained in the most recently filed document. The 3M documents incorporated by reference are:

- annual report on Form 10-K for the fiscal year ended December 31, 2002;
- quarterly reports on Form 10-Q for the quarters ended March 31, June 30, and September 30, 2003;
- current reports on Form 8-K dated March 4, March 26, May 23, August 13, and August 21 and December 15, 2003; and
- description of our common stock contained in our registration statement on Form S-3, filed with the SEC on July 31, 2000 and as amended on August 18, 2000.

All documents filed by 3M with the SEC under Sections 13(a), 13(c), 14, and 15(d) of the Securities Exchange Act of 1934 after the initial filing of the registration statement, or after the date of this registration statement to the date of effectiveness of this registration statement, that contains this prospectus and before the time that our offering is completed (other than current reports furnished under Item 9 and Item 12 of Form 8-K) will also be deemed to be incorporated by reference and will automatically update information in this prospectus.

You can obtain any of the documents incorporated by reference in this prospectus from the SEC through the SEC's web site at the address provided above. Documents incorporated by reference are also available from 3M without charge, excluding any exhibits to those documents, unless the exhibit is specifically incorporated by reference as an exhibit in this proxy statement/prospectus. You can obtain these documents without charge by requesting them in writing or by telephone from 3M at the following addresses:

5

3M Company 3M Center St. Paul, Minnesota 55144 Attention: Investor Relations Telephone number: (651) 733-1110

If you request any incorporated documents from 3M, they will be mailed to you by first class mail, or another equally prompt means, within one business day after receipt of your request.

### USE OF PROCEEDS

All net proceeds from the sale of the common stock covered by this prospectus will go to the selling stockholders who offer and sell their shares. We will not receive any proceeds from the sale of the common stock by the selling stockholders.

### PLAN OF DISTRIBUTION

3M is registering 541,581 shares of our common stock (the "Shares") on behalf of the selling stockholders who may sell the Shares from time to time. The selling stockholders will act independently of 3M in making decisions with respect to the timing, manner and size of each sale. The sales may be made on one or more exchanges or otherwise, at prices and at terms then prevailing or at prices related to the then current market price, or in negotiated transactions. The selling stockholders may effect such transactions by selling the Shares to or through broker-dealers. The Shares may be sold by one or more of, or a combination of, the following:

- a block trade in which a broker-dealer will attempt to sell the Shares as agent but may position and resell a portion of the block as principal to facilitate the transaction.
- purchases by a broker-dealer as principal and resale by such broker-dealer for its account,
- · ordinary brokerage transactions and transactions in which the broker solicits purchasers, and

When selling the Shares, the selling stockholders may enter into hedging transactions. For example, the selling stockholders may:

- enter into transactions after the date of this prospectus involving short sales of the Shares by broker-dealers;
- sell Shares short themselves after the date of this prospectus and deliver the Shares registered hereby to settle such short sales or to close out stock loans incurred in connection with their short positions;
- enter into option or other types of transactions that require the selling stockholder to deliver Shares to a broker-dealer or other person, who will then resell or transfer the

• loan or pledge the Shares to a broker-dealer or other person, who may sell the loaned shares or, in the event of default, sell the pledged shares.

The selling stockholders may negotiate and pay broker-dealers or other persons commissions, discounts or concessions for their services. Broker-dealers or other persons engaged by the selling shareholders may allow other broker-dealers or other persons to participate in resales. However, the selling stockholders and any broker-dealers or such other persons involved in the sale or resale of the Shares may qualify as "underwriters" within the meaning of the Section 2(a)(11) of the Securities Act. In addition, the broker-dealers' or their affiliates' commissions, discounts or concessions may qualify as underwriters' compensation under the Securities Act. If the selling stockholders qualify as "underwriters," they will be subject to the prospectus delivery requirements of Section 5(b)(2) of the Securities Act.

In addition to selling their Shares under this prospectus, the selling stockholders may:

- agree to indemnify any broker-dealer or agent against certain liabilities related to the selling of the Shares, including liabilities arising under the Securities Act;
- transfer their Shares in other ways not involving market makers or established trading markets, including directly by gift, distribution, or other transfer; or
- sell their Shares under Rule 144 of the Securities Act rather than under this prospectus, if the transaction meets the requirements of Rule 144.

The selling stockholders have advised 3M that they have not entered into any agreements, understandings or arrangements with any underwriters or broker-dealers regarding the sale of the Shares and that there is no underwriter or coordinating broker acting in connection with the proposed sale of Shares by selling stockholders.

The Shares may be sold only through registered or licensed brokers or dealers if required under applicable state securities laws. In addition, in certain states the Shares may not be sold unless they have been registered or qualified for sale in the applicable state or an exemption from the registration or qualification requirement is available and is complied with.

Under applicable rules and regulations under the Exchange Act, any person engaged in the distribution of the Shares may not simultaneously engage in market making activities with respect to our common stock for a period of one business day prior to the commencement of such distribution. In addition, each selling stockholder will be subject to applicable provisions of the Exchange Act and the associated rules and regulations under the Exchange Act, including Regulation M, which provisions may limit the timing of purchases and sales of Shares by the selling stockholders. 3M will make copies of this prospectus available to the selling stockholders and has informed them of the need for delivery of copies of this prospectus to purchasers at or prior to the time of any sale of the Shares.

Additional information related to the selling stockholders and the plan of distribution may be provided in one or more supplemental prospectuses.

7

3M will bear all costs, expenses and fees in connection with the registration of the Shares. The selling stockholders will bear all commissions and discounts, if any, attributable to the sales of the Shares.

### SELLING STOCKHOLDERS

The following table sets forth the number of Shares owned by the selling stockholders as of February 5, 2004. In the past three years, none of the selling stockholders has had a material relationship with 3M. No estimate can be given as to the number of Shares that will be held by the selling stockholders after completion of this offering because the selling stockholders may offer all or some of the Shares and because the selling stockholders have advised us that there currently are no agreements, arrangements or understandings with respect to the sale of any of the Shares. The Shares covered by this prospectus may be offered from time to time by the selling stockholders named below.

SELLING STOCKHOLDERS	SHARES BENEFICIALLY OWNED PRIOR TO THE OFFERING (1)	SHARES WHICH MAY BE SOLD PURSUANT TO THIS PROSPECTUS (2)
Stephen R. Berger	74	74
Erin E. Bahr-Kroll	54	54
Joe Blauert	785	785
John N. Capouch	148	148
David K. Cotterman	37	37
Kimberly A. Current	578	578
Robert P. Cushman III	46	46
Thomas Damm	93	93
Duff Davidson	18,871	18,871
DCSI Stock Ownership Trust	100,526	100,526
Michael J. Devine	317	317
Karen Esselman-Batt	68	68

Gemini Investors III, L.P.	187,667	187,667
Terry Groves	131	131
Darin Gunderman	282	282
James L. Hauck	2,163	523
Christopher Heim	18,871	18,871
Robert M. and G. Selena Hiben	1,273	1,273
Robert M. Hiben, Trust Center of America Custodian FBO, IRA	235	235
Randy Allen Hoffman	326	326
Douglas F. Julien	51	51
Steven Kickert	9,057	9,057
Steven Kickert, Dain Rauscher Custodian FBO, IRA	9,813	9,813
Christopher Kosieracki	3,773	3,773
Christopher Kosieracki, Trust Center of America Custodian FBO, IRA	5,660	5,660
Richard A. Kunze	101	101
Jeffrey L. Laurel	50,750	50,750
Laurie McGrath	130	130
Joseph M. McManus	74	74

SELLING STOCKHOLDERS	SHARES BENEFICIALLY OWNED PRIOR TO THE OFFERING (1)	SHARES WHICH MAY BE SOLD PURSUANT TO THIS PROSPECTUS (2)
Chad Miller	19	19
Kevin Nyenhuis	152	152
Janet S. Ostberg	130	130
Barbara J. Pavicic Groven	46	46
Daniel R. Radunz	261	261
Carolina Rayar	43	43
Eric K. Richens	1,373	1,373
Joel Schlachtenhaufen	3,774	3,774
Michael J. Schrader	93	93
Raymond A. Soderberg	3,622	3,622
Starbrook Investments, Inc.	63,849	63,849
Paul K. Steen	654	654
Joel Sternke, Dain Rauscher Custodian FBO, IRA	1,310	1,310
Robert John Stone	134	134

Thomas R. Tolliver	52	52 4,717 42,870	
Carol A. Trusz	4,717		
Gregory A. and Linda R. Twedt, Trustee for the Gregory A. Twedt Revocable Trust U/A	42,870		
Linda R. and Gregory A. Twedt, Trustee for the Linda R. Twedt Revocable Trust U/A	7,893	7,893	
Michael Uchytil	52	52	
Rose Watschke	130	130	
James J. Wille	93	93	
TOTAL	543,221	541,581	

- (1) Represents beneficial ownership of less than one percent of 3M's outstanding capitalization.
- (2) The registration statement of which this prospectus forms a part also shall cover any additional shares of our common stock which become issuable in connection with the Shares by reason of any stock dividend, stock split, recapitalization or other similar transaction effected without the receipt of consideration which results in an increase in the number of 3M's outstanding shares of common stock.

#### LEGAL OPINIONS

Gregg M. Larson, who is our Assistant General Counsel, or another one of our lawyers, will issue an opinion about the validity of the Shares offered in this prospectus, as well as other relevant legal matters. Mr. Larson beneficially owns, or has options to acquire, a number of shares of our common stock, which represents less than 1% of the total outstanding common stock.

#### **EXPERTS**

The financial statements incorporated in this prospectus by reference to 3M Company's Current Report on Form 8-K dated May 23, 2003 have been so incorporated in reliance on the

9

report of PricewaterhouseCoopers LLP, independent accountants, given on the authority of that firm as experts in auditing and accounting.

### **Independent Accountants**

With respect to the unaudited interim financial information of 3M Company for the three-month periods ended March 31, 2003 and 2002, the three- and six-month periods ended June 30, 2003 and 2002 and the three- and nine-month periods ended September 30, 2003 and 2002, incorporated by reference in this prospectus, PricewaterhouseCoopers LLP have reported that they have applied limited procedures in accordance with professional standards for a review of such information. However, their separate reports dated April 21, 2003, July 21, 2003, and October 20, 2003 incorporated by reference herein, state that they did not audit and they do not express an opinion on that unaudited interim financial information. Accordingly, the degree of reliance on their report on such information should be restricted in light of the limited nature of the review procedures applied.

PricewaterhouseCoopers LLP is not subject to the liability provisions of Section 11 of the Securities Act of 1933 for their report on the unaudited interim financial information because that report is not a "report" or a "part" of the registration statement prepared or certified by PricewaterhouseCoopers LLP within the meaning of Sections 7 and 11 of the Securities Act.

We have not authorized any person to make a statement that differs from what is in this prospectus. If any person does make a statement that differs from what is in this prospectus, you should not rely on it. This prospectus is not an offer to sell, nor is it seeking an offer to buy, these securities in any state in which the offer or sale is not permitted. The information in this prospectus is complete and accurate as of its date, but the information may change after that date.

### **3M COMPANY**

### 541,581 SHARES OF COMMON STOCK

PROSPECTUS \_\_\_\_\_, 2004

11

### PART II

### INFORMATION NOT REQUIRED IN PROSPECTUS

Item 14. Other Expenses of Issuance and Distribution.

The following table sets forth the estimated expenses payable by the registrant in connection with the offering described in this registration statement.

SEC registration fee	\$ 5,393
Legal fees and expenses	15,000
Auditors' fees and expenses	10,000
Printing expenses	2,300
Miscellaneous expenses	2,000
TOTAL	\$ 34,693

Item 15. Indemnification of Directors and Officers.

Our Certificate of Incorporation eliminates the liability of directors to the fullest extent permitted by the General Corporation Law of the State of Delaware, which currently permits a corporation to eliminate the liability of a director for monetary damages for breach of the duty of care, subject to appropriate stockholder approval. In addition, our Bylaws contain provisions entitling directors, officers, and employees to indemnification to the fullest extent permitted by current Delaware law.

Item 16. Exhibits.

Exhibit Description

- 2.1 Amended Agreement and Plan of Merger dated as of January 14, 2004 among 3M Company, Steeler Merger Corporation, Steeler Merger LLC, and HighJump Software, Inc.
- 5.1 Opinion of Gregg M. Larson, Assistant General Counsel of 3M as to the legality of the securities being registered and consent to the use of the opinion in this registration statement
- 15 Awareness Letter of PricewaterhouseCoopers LLP (regarding interim financial information)
- 23.1 Consent of Gregg M. Larson (included in opinion filed as Exhibit 5.1)
- 23.2 Consent of PricewaterhouseCoopers LLP
- 24.1 Powers of Attorney

#### Item 17. Undertakings.

- (a) The undersigned Registrant hereby undertakes:
  - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
    - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act of 1933;

II-1

- (ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement, or the most recent post-effective amendment thereof, which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement;
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement;

provided, however, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the registration statement is on Form S-3, Form S-8 or Form F-3, and the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by the Registrant pursuant to Section 13 or 15(d) of the Exchange Act of 1934 that are incorporated by reference in the registration statement.

- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof;
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering;
- (b) The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act of 1934 that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

II-2

### SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the Registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-3 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of St. Paul, and State of Minnesota, on the 6<sup>th</sup> day of February, 2004.

3M COMPANY

By: /s/ Gregg M. Larson

Name: Gregg M. Larson

Title: Assistant General Counsel and Secretary

Signature	Title
*	Chairman of the Board, Chief Executive Officer and Director
W. James McNerney, Jr	
*	Senior Vice President and Chief Financial Officer
Patrick D. Campbell	(Principal Financial and Accounting Officer)
*	Vice President and Controller
Ronald G. Nelson	
*	Director
Linda G. Alvarado	
*	Director
Edward A. Brennan	
*	Director
Vance D. Coffman	
*	Director
Michael L. Eskew	
*	Director
Edward M. Liddy	
*	Director
Robert S. Morrison	
*	Director
Aulana L. Peters	
	II-3
*	Director
Rozanne L. Ridgway	
*	Director
Kevin W. Sharer	
*	Director
Louis W. Sullivan	
: /s/ Gregg M. Larson	
Gregg M. Larson Attorney-in-fact Date: February 6, 2004	

### INDEX TO EXHIBITS

Exhibit Exhibit Title

2.1 Amended Agreement and Plan of Merger dated as of January 14, 2004 among 3M Company, Steeler Merger Corporation, Steeler Merger LLC, and HighJump Software, Inc.

The registrant hereby agrees to furnish supplementally to the Commission upon request a copy of any omitted exhibit or schedule contained in the agreement.

- 5.1 Opinion of Gregg M. Larson, Assistant General Counsel of 3M as to the legality of the securities being registered and consent to the use of the opinion in this registration statement.
- 15 Awareness Letter of PricewaterhouseCoopers LLP (regarding interim financial information):
- 23.1 Consent of Gregg M. Larson (included in opinion filed as Exhibit 5.1)
- 23.2 Consent of PricewaterhouseCoopers LLP
- 24.1 Powers of Attorney

\_ \_\_\_\_\_\_

### AMENDED

### AGREEMENT AND PLAN OF MERGER

BY AND AMONG

3M COMPANY,

STEELER MERGER CORPORATION,

STEELER MERGER LLC

AND

HIGHJUMP SOFTWARE, INC.

Dated as of January 14, 2004

\_ \_\_\_\_\_\_\_

### TABLE OF CONTENTS

	PAGE
ARTICLE I TH	E MERGER2
1.01.	Merger2
1.02.	Effective Time of the Merger2
1.03.	Articles of Incorporation and Bylaws of the Surviving
	Corporation
1.04.	Board of Directors and Officers of the Surviving
1.05.	Corporation
1.05.	Conversion of Shares; Merger Consideration
1.06.	Additional Consideration
1.07.	Appraisal Rights
1.00.	Payment for Shares
1.10.	Distribution of Cash Flow Holdback
1.11.	No Further Rights or Transfers8
1.12.	Tax Treatment8
1.12.	Tan Treatment
ARTICLE II C	CLOSING9
2.01.	Generally9
2.02.	Deliveries at the Closing9
2.02.	belivered at the closing
ARTICLE III	REPRESENTATIONS AND WARRANTIES OF THE COMPANY9
3.01.	Organization, Standing, Qualification9
3.02.	Capitalization10
3.03.	Authorization and Execution10
3.04.	No Conflicts11
3.05.	Financial Statements11

3.	.06.	Absence of Certain Changes or Events12
3.	.07.	Tax Matters
3.	.08.	Real Property14
3	09.	Title to Properties
	10.	Material Contracts
	11.	Intellectual Property
	12.	Litigation20
3.	13.	Permits, Licenses, Authorizations; Compliance with Laws20
3.	14.	Retirement and Benefit Plans20
3.	15.	Employees
	16.	Environmental Matters
		Insurance
	17.	
	18.	Customer Relationships25
3.	19.	Product Warranty
3.	20.	Conditions Affecting Business
3.	21.	No Brokers or Finders
		-i-
		-1-
3.	22.	Disclosure
3.	23.	Sole Representations and Warranties
		1
λοπτατπ	ייות עד	PRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUBSIDIARY27
VVIICPE	TA KE	TRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUBSTITIARY2/
	01.	Organization, Good Standing, Equity Ownership27
4.	02.	Authorization and Execution27
4.	.03.	No Conflicts27
4	04.	Litigation28
	.05.	SEC Filings; Financial Statements
	.06.	No Brokers or Finders
	.07.	Interim Operations of Merger Subsidiary28
4.	.08.	Sole Representations and Warranties29
ARTICLE	V CONI	DUCT AND TRANSACTIONS PRIOR TO THE EFFECTIVE TIME29
5	01.	Operation of Business of the Company Until Effective Time29
	.02.	Shareholder Approval30
5.	.03.	No Shopping
5.	.04.	Access to Information31
5.	.05.	Confidentiality Agreement31
5.	06.	Deposit of Escrowed Funds31
	07.	Reasonable Efforts
		Amendment of Disclosure Schedules
	.80.	
	.09.	Employment Agreements32
5.	10.	Incentive Programs32
5.	11.	Retention Program
5.	12.	Registration and Listing of Shares
	13.	Restrictive Legend on Shares32
	14.	Taxes on Transfer
5.	15.	Conditions to Parent's Obligation to File Registration
		Statement33
5.	16.	Investment Representations
5.	17.	Stock Ownership Plan
5.	18.	Non-Solicitation Agreements34
5.	19.	Redemption of Shares34
	20.	Resignation of Employees
٥.	. 20.	Resignation of Employees
ADDICID	117 001	NDITIONS PRECEDENT
ARTICLE	VI COI	NDITIONS PRECEDENT
6.	01.	Conditions to the Obligations of Parent and Merger Subsidiary34
6.	02.	Conditions to the Obligations of the Company
ARTICLE	VII CO	ONDUCT AND TRANSACTIONS AFTER THE EFFECTIVE TIME
MILLOHE	V 1 1 C	ondoer and inducations arise the streetive right
-	0.1	
	.01.	Indemnification
7.	.02.	Directors and Officers Liability Insurance
7.	.03.	Confidentiality37
		-ii-
7	0.4	Employment Fellowing Effective Min-
	04.	Employment Following Effective Time
7.	.05.	Completion of Merger between New LLC and the Company38
ARTICLE	VIII	TERMINATION AND ABANDONMENT
ρ	01.	Generally
	.02.	Procedure and Effect of Termination and Abandonment39
8.	.∪∠.	riocedure and firect of Termination and Abandonment
ARTICLE	IX SH	AREHOLDER REPRESENTATIVE39
9.	01.	Designation
	.02.	Authority
۶.	· · ·	

9.03. 9.04.	Resignation
9.05.	Exculpation and Indemnification40
ARTICLE X INDEMNIFICATION	
10.01. 10.02. 10.03. 10.04. 10.05. 10.06. 10.07. 10.08. 10.09.	Indemnification by the Shareholders
ARTICLE XI MISCELLANEOUS PROVISIONS47	
11.01. 11.02. 11.03. 11.04. 11.05. 11.06. 11.07. 11.08. 11.09. 11.10. 11.11. 11.12. 11.13. 11.14. 11.15.	Survival of Representations, Warranties and Covenants of the Company, Parent and Merger Subsidiary
Exhibit 2.1	
11.16. 11.17. 11.18. 11.19. 11.20.	Enforcement
EXHIBITS	
Exhibit B - Exhibit C -	Form of Escrow Agreement Key Employee Matters Form of Incentive Program 1 Form of Incentive Program 2

Exhibit E - Form of Retention Program

-iv-

EXECUTION COPY

subsidiary of Parent ("Merger Subsidiary"), STEELER MERGER LLC, a Delaware limited liability company and wholly owned subsidiary of Parent ("New LLC"), and HIGHJUMP SOFTWARE, INC., a Minnesota corporation (the "Company").

#### RECTTALS

- A. Parent, Merger Subsidiary, the Company and New LLC previously executed and delivered an Agreement and Plan of Merger dated as of December 23, 2003 (the "Original Merger Agreement") and desire to amend and restate the Original Merger Agreement in order to make certain technical changes, the primary purpose of which is to facilitate compliance with United States federal securities laws.
- B. The Boards of Directors of Parent, Merger Subsidiary and the Company have determined that it is advisable and in the best interests of their respective shareholders that the parties consummate the business combination provided for herein in which Merger Subsidiary will merge with and into the Company with the Company being the surviving corporation in the merger (the "Merger").
- C. Immediately following the Merger, and in any event no later than the next business day following the Merger, Parent will cause the Company to merge with and into New LLC.
- D. Parent, as the sole shareholder of Merger Subsidiary, has approved this Agreement, the Merger and the transactions contemplated by this Agreement pursuant to actions taken in accordance with the requirements of the Minnesota Business Corporation Act (the "MBCA") and the Delaware General Corporation Law (the "DGCL").
- E. The Board of Directors of the Company has determined that the Merger is advisable and in the best interests of the Company and its shareholders and has directed that this Agreement be submitted to a vote of the shareholders of the Company (the "Shareholders").
- F. The Merger, immediately followed by the merger of the Company with and into New LLC, is intended to be an integrated transaction that qualifies as a reorganization under the provisions of Section 368(a) of the United States Internal Revenue Code of 1986, as amended (the "Code")
- G. Reference is made to Section 11.20 hereof, which lists the Sections within this Agreement where each capitalized term used herein is defined.

### AGREEMENT

NOW, THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements in this Agreement, the parties agree as follows:

### ARTICLE I

### THE MERGER

- 1.01. MERGER. At the Effective Time (as hereinafter defined) and in accordance with the terms and subject to the conditions set forth in this Agreement and the MBCA, the Merger Subsidiary shall be merged with and into the Company , the separate corporate existence of the Merger Subsidiary shall thereupon cease, and the Company shall be the surviving corporation in the Merger (sometimes hereinafter referred to as the "Surviving Corporation"), and the Surviving Corporation shall thereupon and thereafter possess all the rights and obligations of each of the constituent corporations in accordance with the MBCA. At the Effective Time, the Merger shall have the other effects provided in the applicable provisions of the MBCA. As set forth in Section 7.05 below, as soon as possible thereafter, but not later than the close of business on the first business day immediately following the Effective Time, the Surviving Corporation, as part of a single plan to which the Merger is a part, shall be merged with and into New LLC in accordance with the provisions of Minnesota law and the Delaware General Corporation Law, such that, after giving effect to such merger, New LLC shall ultimately be the surviving entity in the Merger, and, where appropriate, the term Surviving Corporation shall refer to New LLC.
- 1.02. EFFECTIVE TIME OF THE MERGER. Concurrently with the Closing (as hereinafter defined), the parties hereto shall cause the Merger to be consummated by filing articles of merger (the "Articles of Merger"), in such form as required by, and executed in accordance with, the applicable provisions of the MBCA, with the Secretary of State of the State of Minnesota and shall make all other filings or recordings required under the MBCA. The Merger shall become effective upon the filing of the Articles of Merger with the Secretary of State of the State of Minnesota in accordance with the MBCA or such later date or time as the parties shall agree and specify in the Articles of Merger. The term "Effective Time" shall mean the date and time the Merger becomes effective in accordance with the MBCA.

- 1.03. ARTICLES OF INCORPORATION AND BYLAWS OF THE SURVIVING CORPORATION. The Articles of Incorporation of the Company in effect immediately prior to the Effective Time shall be the Articles of Incorporation of the Surviving Corporation, until amended in accordance with the laws of the State of Minnesota and such Articles of Incorporation. The Bylaws of the Company in effect immediately prior to the Effective Time shall be the Bylaws of the Surviving Corporation, until further amended in accordance with the laws of the State of Minnesota, the Articles of Incorporation of the Surviving Corporation and such Bylaws.
- 1.04. BOARD OF DIRECTORS AND OFFICERS OF THE SURVIVING CORPORATION. The directors of Merger Subsidiary immediately prior to the Effective Time shall be the directors of the Surviving Corporation, each of such directors to hold office, subject to the applicable provisions of the Articles of Incorporation and Bylaws of the Surviving Corporation, until the

-2-

expiration of the term for which such director was elected and until his or her successor is elected and has qualified or as otherwise provided in the Articles of Incorporation or Bylaws of the Surviving Corporation. The officers of the Company immediately prior to the Effective Time shall be the officers of the Surviving Corporation until their respective successors are chosen and have qualified or as otherwise provided in the Bylaws of the Surviving Corporation.

#### 1.05. CONVERSION OF SHARES; MERGER CONSIDERATION.

- (a) Parent shall pay to the holders of convertible preferred stock, preferred stock, common stock, warrants, stock options, and any other securities or ownership rights in Company, an aggregate merger consideration of \$68,000,000. At the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Subsidiary, the Company or the holders of any of the following securities, \$68,000,000 in aggregate merger consideration shall be paid by Parent as follows:
  - (i) each holder of shares of common stock of the Company issued and outstanding immediately prior to the Effective Time (other than with respect to any shares of Company Common Stock with respect to which such holders thereof have perfected appraisal rights pursuant to Section 1.07, and any shares redeemed by the Company pursuant to Section 5.19) shall receive the consideration set forth across from such holder's name on Schedule 1.05, as such schedule shall be updated by the parties hereto immediately prior to the Effective Time to reflect any exercise of outstanding stock options or redemption of shares between the date of the Original Merger Agreement and the Effective Time, in shares of common stock of Parent ("Parent Common Stock") based upon the Parent Share Value (as hereinafter defined) as of the day before the Closing Date;
  - (ii) each holder of shares of preferred stock of the Company issued and outstanding immediately prior to the Effective Time (other than with respect to any shares of preferred stock of the Company which such holders thereof have perfected appraisal rights pursuant to Section 1.07) and warrants to acquire shares of preferred stock of the Company and each holder of warrants outstanding immediately prior to the Effective Time (collectively, the "Warrants") shall receive consideration in the amounts set forth across from such holder's name on Schedule 1.05, as such schedule shall be updated by the parties hereto immediately prior to the Effective Time to reflect any exercise of outstanding stock options or redemption of shares between the date of the Original Merger Agreement and the Effective Time, in cash and/or shares of Parent Common Stock based upon the Parent Share Value (as hereinafter defined) as of the day before the Closing Date.
  - (iii) each option to acquire shares of common stock of the Company outstanding immediately prior to the Effective Time shall be cancelled in exchange for a cash payment described in Section 1.08 and Schedule 1.08, as such schedule shall be updated by the parties hereto immediately prior to the Effective Time to reflect any exercise, expiration or termination of outstanding stock

the Effective Time.

- (iv) all other securities of the Company shall be cancelled and shall have no further force and effect, and the holders of such securities shall have no right to receive any portion of the consideration payable by Parent under this Agreement; and
- (v) each share of common stock, \$0.01 par value per share, of the Merger Subsidiary issued and outstanding immediately prior to the Effective Time shall be converted into and exchanged for one validly issued, fully paid and nonassessable share of common stock, \$0.01 par value, of the Surviving Corporation.
- (b) If, during the period between the date hereof and the Effective Time, any change in the capital stock of Parent shall occur by reason of reclassification, recapitalization, stock split or combination, exchange or readjustment of shares, or any stock dividend thereon with a record date during such period or any similar event, the amounts of shares of Parent Common Stock issuable pursuant to this Agreement shall be appropriately adjusted.
- (c) Notwithstanding any other provision in this Agreement to the contrary, the sum of (i) the total aggregate amount of cash that will be paid in the Merger pursuant to this Section 1.05, (ii) any cash amounts to be paid in exchange for fractional shares, (iii) any cash amounts to be paid for Dissenting Shares pursuant to Section 1.07, (iv) any other amounts paid by Parent or the Company to or on behalf of any shareholder of the Company in connection with the sale or other disposition of any shares of capital stock of the Company in connection with the Merger for purpose of Treasury Regulation Section 1.368-1(e), and (v) the amount of any extraordinary dividend (including any cash distributions and payments made pursuant to Section 1.06) distributed and paid by the Company prior to and in connection with the Merger (the sum of these amounts, the "Aggregate Cash Amount") shall not exceed 50% of the Aggregate Consideration (as defined in this Section). The "Aggregate Consideration" shall equal the sum of (i) the Aggregate Cash Amount, plus (ii) the number of shares of Parent Common Stock to be issued in the Merger times the average of the highest and lowest quoted trading price of Parent Common Stock on the date of the Effective Time. If the Aggregate Cash Amount would exceed 50% of the Aggregate Consideration, the number of shares of Parent Common Stock that would be issued to holders of capital stock of the Company shall be increased and the amount of cash that would be issued to the holders of capital stock of the Company shall be decreased so that the Aggregate Cash Amount does not exceed 50% of the Aggregate Consideration. The foregoing adjustments shall be applied in a manner such that after such adjustments the sum of (i) the Aggregate Cash Amount and (ii) the product of the number of shares of Parent Common Stock to be issued in the Merger multiplied by the Parent Share Value as of the day immediately prior to the Closing Date, shall be equal to such sum absent such adjustments.

-4-

(d) For purposes of this Agreement, "Parent Share Value" means the average of the daily closing prices of a share of Parent Common Stock (calculated to the nearest 0.0001) on The New York Stock Exchange, as reported in the New York Edition of the Wall Street Journal, for the ten (10) consecutive trading days immediately preceding and including the applicable date.

### 1.06. ADDITIONAL CONSIDERATION.

- (a) Immediately prior to the Effective Time, those persons whose names are listed on Schedule 1.06(a), including certain holders of shares of capital stock of the Company, holders of Stock Options (as hereinafter defined) and an adviser to the Company, shall be entitled to receive a cash distribution from the Company in the amounts set forth across from such persons' names on Schedule 1.06(a), as such schedule shall be updated by the parties hereto immediately prior to the Effective Time to reflect any exercise, expiration or termination of outstanding stock options or redemption of shares between the date of the Original Merger Agreement and the Effective Time. Prior to the Closing, the Company shall prepare and deliver to Parent a balance sheet of the Company as of December 9, 2003 (the "Closing Balance Sheet"). The Cash balance on the Closing Balance Sheet up to \$14,000,000 (less any required tax withholding) shall be distributed as set forth in Schedule 1.06(a) until working capital equals \$500,000 (the amount payable in accordance with the foregoing being referred to herein as "Closing Cash Consideration").
- (b) If the Closing Balance Sheet shows a Cash Balance greater than the sum of \$14,000,000\$ and the aggregate amount of legal and

accounting fees of the Company incurred in connection with the Merger (the "Transaction Fees"), then the Cash Balance in excess of the sum of \$14,000,000 and the Transaction Fees shall be distributed pro rata to the holders of capital stock in the Company until working capital, after deducting (i) the sum of the amounts distributed under Sections 1.06(a) and 1.06(b) and (ii) the Transaction Fees, equals \$500,000 (the amount payable in accordance with the foregoing being referred to herein as "Additional Closing Cash Consideration").

1.07. APPRAISAL RIGHTS. Notwithstanding anything in this Agreement to the contrary, each share (if any) of capital stock of the Company issued and outstanding immediately before the Effective Time for which the holder has properly exercised and perfected such holder's demand for appraisal rights under Sections 302A.471 and 302A.473 of the MBCA (each a "Dissenting Share") shall not be converted into the right to receive its portion of the consideration specified in Sections 1.05 and 1.06 at or after the Effective Time unless and until the holder of such shares withdraws such holder's demand for appraisal rights or becomes ineligible for appraisal rights. If any such holder fails to perfect (or otherwise loses) any such appraisal rights, then each such share of such holder shall be treated as a share that had been converted as of the Effective Time into the right to receive its portion of the merger consideration specified in Sections 1.05 and 1.06, without interest. The Company shall give prompt notice to Parent of each demand received by the Company for payment of fair value of Common Stock, and Parent shall have the right to participate in negotiations and proceedings regarding each such demand. The Company shall not, except with prior written consent of Parent, settle or make any payment regarding any such demand. Each person holding of record

-5-

or beneficially owning Dissenting Shares who becomes entitled under Sections 302A.471 and 302A.473 of the MBCA and this Section 1.07 to payment of the fair value of such Dissenting Shares (and any other payments required by Sections 302A.471 and 302A.473 of the MBCA) shall receive payment therefor from the Surviving Corporation.

1.08. STOCK OPTIONS. At or immediately prior to the Effective Time, each then outstanding Stock Option (as hereinafter defined) shall be canceled by the Company in exchange for a cash payment by the Company to the holder of such Stock Option equal to the difference between (a) the per share amount based upon (i) the total consideration to be paid to the holders of shares of common stock, Stock Options and Warrants and preferred stock, on an as converted basis, of the Company issued and outstanding as of immediately prior to the Effective Time (the "Fully Diluted Shares"), divided by (ii) the number of Fully Diluted Shares, and (b) the per share exercise price under the applicable option agreement multiplied by the number of Stock Options held. The aggregate amount payable to the holders of Stock Options under this Section 1.08 is set forth on Schedule 1.08 (which schedule shall be updated by the parties hereto immediately prior to the Effective Time to reflect any exercise, expiration or termination of outstanding stock options between the date of the Original Merger Agreement and the Effective Time) and is referred to herein as the "Option Settlement Amount." Each such cancellation and payment shall occur pursuant to the terms and conditions established for such cancellation in the applicable stock option agreement and the Company's Second Amended and Restated 2000 Stock Option Plan (the "2000 Plan"). Parent shall fund the total amount set forth in Schedule 1.08 in the column "Paid Out of Merger Consideration" at or immediately prior to the Effective Time through a cash payment to the Company, after which the Company shall pay to each holder of Stock Options an amount calculated in accordance with this Section 1.08 and as set forth in Schedule 1.08 (less any applicable withholding Taxes). Payment of the Option Settlement Amount in this manner shall be considered part of and to satisfy in part Parent's obligation to pay the aggregate merger consideration of \$68,000,000 pursuant to Section 1.05. For purposes of this Agreement, "Stock Options" means the options to purchase shares of common stock of the Company outstanding immediately prior to the Effective Time.

### 1.09. PAYMENT FOR SHARES.

- (a) At and after the Effective Time, each holder of a certificate or certificates representing shares of Common Stock canceled and extinguished at the Effective Time may surrender such certificate or certificates to the Parent, to effect the exchange of such certificate or certificates on such holder's behalf. Until so surrendered and exchanged, each outstanding certificate which, prior to the Effective Time, represented shares of Common Stock shall be deemed to represent and evidence only the right to receive the portion of the merger consideration to be paid therefor as set forth in Sections 1.05 and 1.06 and until such surrender and exchange, no cash shall be paid to the holder of such outstanding certificate in respect thereof.
- (b) The Parent shall deliver to the Escrow Agent at the Effective Time (i) \$3,400,000 (the "Indemnification Escrow Amount") and (ii) \$500,000 (the "Cash Flow Holdback"), to be held by the Escrow Agent subject to the terms and conditions of this Agreement and the

-6-

amounts otherwise payable to the holders of common stock and preferred stock of the Company pursuant to Section 1.05, as described in Schedule 1.05.

- (c) If payment is to be made to a person other than the person in whose name the certificate surrendered in exchange therefor is registered, it shall be a condition to such payment that the certificate so surrendered shall be properly endorsed and otherwise in proper form for transfer, and that the person requesting such payment shall pay to the Company any transfer and other taxes required by reason of such payment in any name other than that of the registered holder of the certificate surrendered or shall have established to the satisfaction of the Company that such tax either has been paid or is not payable.
- (d) No interest shall accrue or be payable with respect to any, amounts which a holder of shares of capital stock of the Company, Stock Options or Warrants shall be so entitled to receive. Parent and the Surviving Corporation shall be authorized to pay the consideration attributable to any certificate previously issued which has been lost or destroyed, upon receipt of satisfactory evidence of ownership of the shares of securities represented thereby and of appropriate indemnification.

#### 1.10. DISTRIBUTION OF CASH FLOW HOLDBACK.

- (a) Within 210 days following the Closing Date, Parent shall deliver to the Stockholders' Representatives a statement of cash flows (the "PROPOSED FINAL CASH FLOW STATEMENT") reflecting Parent's determination of the amount of cash flows generated by the Company from continuing operations of the Company during the six-month period immediately following the Closing Date and ending on the six-month anniversary of the Closing Date, without taking into account any general corporate allocations of Parent to the Company that do not specifically relate to the business of the Company, all in accordance with Schedule 1.10 hereto (the "PROPOSED CASH FLOW AMOUNT"), together with the books, records, work papers and similar documentation used in Parent's calculation of the Proposed Cash Flow Amount.
- (b) After receipt of the Proposed Final Cash Flow Statement, the Stockholders' Representatives shall have thirty (30) days to review the Proposed Final Cash Flow Statement, together with the books, records, workpapers and similar documentation used by Parent in its preparation and calculation of the Proposed Cash Flow Amount, and shall have access to the Company's books, records and personnel upon reasonable notice and during normal business hours in conducting such review. The Proposed Final Cash Flow Statement and the Proposed Cash Flow Amount shall become final and binding (in their final and binding form, after resolution of any disputes hereunder, the "FINAL CASH FLOW STATEMENT" and the "FINAL CASH FLOW AMOUNT", respectively) on the thirtieth day following receipt thereof by the Stockholders' Representatives unless the Stockholders' Representatives give written notice of their disagreement (a "NOTICE OF DISAGREEMENT") to Parent prior to such date. Any Notice of Disagreement shall specify in reasonable detail the nature and amount of any disagreement so asserted. If a timely Notice of Disagreement is delivered by the Stockholders' Representatives, then the parties shall work in good faith to resolve such

-7-

disagreement, and if the parties cannot resolve such disagreement within thirty (30) days after delivery by the Stockholders' Representatives of the Notice of Disagreement to the Proposed Final Cash Flow Statement and Proposed Cash Flow Amount, such disagreement shall be referred to a nationally recognized "Big Four" independent accounting firm mutually satisfactory to Parent and the Stockholders' Representatives (the "REVIEWING PARTY"), which shall be directed to resolve such disagreement within thirty (30) days thereafter, and whose decision shall be final and binding on all parties.

- (c) The fees and expenses of the Reviewing Party retained pursuant to this Section 1.10 shall be paid as follows:
  - (i) if the Reviewing Party determines that the Final Cash Flow Amount is greater than the Proposed Cash Flow  $\,$

Amount, all of the fees and expenses of the Reviewing Party shall be borne by Parent; and

- (ii) if the Reviewing Party determines that the Final Cash Flow Amount is equal to or less than the Proposed Cash Flow Amount, all of the fees and expenses of the Reviewing Party shall be disbursed from the Indemnification Escrow Amount.
- (d) If the Final Cash Flow Amount is equal to or greater than \$1.00, Parent shall cause the Escrow Agent to distribute the Cash Flow Holdback to the Stockholders' Representatives, for further payment pro rata to the former holders of capital stock of the Company, within three business days following the date on which the Final Cash Flow Statement is determined. If the Final Cash Flow Amount is less than zero, Parent shall cause the Escrow Agent to distribute (i) the Cash Flow Holdback, plus (ii) the amount of the Final Cash Flow Amount, to the Stockholders' Representatives for further payment pro rata to the former holders of capital stock of the Company. Any remaining Cash Flow Holdback shall be distributed to the Parent within three business days following the date on which the Final Cash Flow Statement is determined.
- 1.11. NO FURTHER RIGHTS OR TRANSFERS. At the Effective Time, all shares of capital stock of the Company issued and outstanding immediately prior to the Effective Time shall be canceled and cease to exist, and each holder of a certificate or certificates that represented shares of capital stock issued and outstanding immediately prior to the Effective Time shall cease to have any rights as a Shareholder with respect to the shares of capital stock represented by such certificate or certificates, except for the right to surrender such certificate or certificates in exchange for the payment provided pursuant to Sections 1.05 and 1.06 hereof or to preserve and perfect such holder's right to receive payment for such holder's shares pursuant to Sections 302A.471 and 302A.473 of the MBCA and Section 1.07 hereof if such holder has validly exercised and not withdrawn or lost such right, and no transfer of shares of capital stock issued and outstanding immediately prior to the Effective Time shall be made on the stock transfer books of the Surviving Corporation.
- 1.12. TAX TREATMENT. The parties intend that the Merger, considered together with the merger contemplated in Section 7.05 below, be treated as a reorganization described in

-8-

Section 368(a) of the Code. The parties to this Agreement shall not take a position on any Tax Returns inconsistent with such treatment unless otherwise required by law.

### ARTICLE II

### CLOSING

- 2.01. GENERALLY. Subject to the fulfillment or waiver of the conditions precedent set forth in Article VI hereof and the termination provisions set forth in Article VIII hereof, the closing (the "Closing") of the transactions contemplated hereby shall occur on the last business day of the calendar month during which the first business day following the day on which the fulfillment or waiver of the conditions precedent set forth in Article VI occurs, or on such other date as Parent and the Company may mutually agree (the "Closing Date"). The Closing shall be held at the offices of Dorsey & Whitney LLP, 50 South Sixth Street, Suite 1500, Minneapolis, Minnesota 55402, or at such other place as Parent and the Company may mutually agree.
- 2.02. DELIVERIES AT THE CLOSING. Subject to the provisions of Articles VI and VIII hereof, at the Closing:
  - (a) There shall be delivered to Parent, Merger Subsidiary and the Company the certificates and other documents and instruments as contemplated under Article V or VI hereof; and
  - (b) The Company and Merger Subsidiary shall cause the Articles of Merger to be filed as provided in Section 1.02 hereof and shall take any and all other lawful actions and do any and all other lawful things necessary to cause the Merger to become effective.

### ARTICLE III

### REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Parent and Merger Subsidiary that, except as otherwise set forth in the disclosure schedules attached hereto (the "Disclosure Schedules" and each a "Disclosure Schedule"), the statements in this Article III are true and correct as of the date of the Original Merger Agreement (for purposes of this Article III, "the date of this Agreement" shall

3.01. ORGANIZATION, STANDING, QUALIFICATION. The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Minnesota and has the requisite corporate power and authority to own, lease and operate all of its properties and assets and to carry on its business as it is now being conducted. The Company is duly qualified as a foreign corporation to do business, and is in good standing, in each jurisdiction where the character of its properties owned, operated or leased, or the nature of its activities, makes such qualification necessary, except such jurisdictions where failure to be so qualified would not, individually or in the aggregate, have a material adverse effect upon the business, operations, properties or financial condition of the Company ("Material Adverse Effect");

-9-

provided, however, that wherever such term is used in this Agreement, a Material Adverse Effect will not be deemed to have occurred if the change, circumstance, event, effect or state of facts results primarily from (a) a change in general economic conditions, (b) a change in general business conditions in the Company's industry which does not disproportionately affect the Company or (c) the disclosure to the public, or pendency, of the Merger and the transactions contemplated thereby. The copies of the Articles of Incorporation and Bylaws of the Company that have been made available to Parent are complete and correct as of the date of this Agreement, and the minute book of the Company that has been made available to Parent is complete in all material respects and accurately reflects all material action taken prior to the date of this Agreement by the Board of Directors of the Company and the Shareholders.

#### 3.02. CAPITALIZATION.

- (a) The authorized capital stock of the Company consists of 75,000,000 shares, of which 8,700,000 shares are designated Series A Convertible Preferred Stock, 7,300,000 shares are designated Series B Convertible Preferred Stock, 50,000,000 shares are designated Common Stock and 9,000,000 shares are undesignated shares of capital stock. At the date hereof, the issued and outstanding capital stock of the Company consists of 8,646,999 shares of Series A Convertible Preferred Stock, 7,185,116 shares of Series B Convertible Preferred Stock and 20,284,427 shares of Common Stock. Disclosure Schedule 3.02(a) contains a complete and correct list of all record owners of shares of capital stock of the Company and the number of shares owned by each such person, in each case at the date hereof. All of the outstanding shares of capital stock of the Company are validly issued, fully paid and nonassessable. At the date hereof, the Company has no other issued or outstanding shares of capital stock.
- (b) Disclosure Schedule 3.02(b) contains a complete and correct list of all outstanding and unexercised Stock Options under the Company's 2000 Stock Option Plan, as amended (the "2000 Plan"), specifying the name of each optionee, the date on which each Stock Option was granted, the number of shares of Common Stock that may be purchased pursuant to each Stock Option, the exercise price at which such shares may be purchased, the vesting period for each Stock Option and the expiration date of each Stock Option. Except for the Stock Options and the Warrants and as otherwise set forth on Disclosure Schedule 3.02(b), there are no outstanding subscriptions, options, warrants, calls or other agreements or commitments by which the Company is bound in respect of the capital stock of the Company, whether issued or unissued, and no outstanding securities convertible into or exchangeable for any such capital stock.
- (c) The Company does not own, directly or indirectly, any capital stock or other equity interest in (or any rights to acquire any capital stock of or other equity interest in) any corporation, partnership, joint venture or other entity.
- 3.03. AUTHORIZATION AND EXECUTION. The Company has the corporate power and authority to execute and deliver this Agreement and, subject to obtaining the Shareholder Approval, to consummate the transactions contemplated hereby. The execution, delivery and performance of this Agreement have been duly and effectively authorized by the Board of Directors of the Company, and no further corporate action of the Company, other than the

-10-

Shareholder Approval, is necessary to consummate the transactions contemplated hereby. This Agreement constitutes a legal, valid and binding obligation of the Company, enforceable against it in accordance with its terms, except to the extent that enforceability may be limited by applicable bankruptcy, insolvency or similar laws affecting the enforcement of creditors' rights generally, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a court of law or equity).

3.04. NO CONFLICTS. Subject to obtaining the Shareholder Approval and except as set forth in Disclosure Schedule 3.04, neither the execution and delivery of this Agreement by the Company, nor the consummation by the Company of the transactions contemplated hereby, will (i) conflict with or result in a breach of the Articles of Incorporation or Bylaws, as currently in effect, of the Company, or (ii) except for the filing of the Articles of Merger as required by the MBCA and any filing required pursuant to the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder (the "HSR Act"), require any filing with, or consent or approval of, any governmental authority having jurisdiction over any of the business or assets of the Company, or (iii) violate any material statute, regulation, injunction, judgment or order to which the Company is subject, or (iv) result in a material breach of, or constitute a material default or an event which, with the passage of time or the giving of notice or both would constitute a material default, give rise to a right of termination, cancellation or acceleration, create any entitlement to any payment or benefit, require the consent of any third party or result in the creation of any lien on the assets of the Company under any Material Contract (as defined in Section 3.10).

3.05. FINANCIAL STATEMENTS. The Company has previously delivered to Parent its (i) audited financial statements for the twelve months ended March 31, 2003, and (ii) unaudited financial statements for the six-month period ended September 30, 2003. All of the foregoing financial statements are hereinafter collectively referred to as the "Financial Statements." The audited Financial Statements were prepared in accordance with generally accepted accounting principles applied on a consistent basis throughout the periods involved (except as may be indicated in the notes thereto) and fairly present, in all material respects, the financial position of the Company as at the respective dates thereof and the results of operations and cash flows of the Company for the periods indicated. The unaudited Financial Statements were prepared by the Company in accordance with generally accepted accounting principles and consistent with the past practices of the Company and fairly present, in all material respects, the financial position of the Company as at the respective dates thereof and the results of operations for the periods indicated, except that all unaudited Financial Statements are subject to normal year-end adjustments and do not contain all footnote disclosures required by generally accepted accounting principles.

Other than as and to the extent (w) disclosed or reserved against in the balance sheet (the "Base Balance Sheet") dated as of September 30, 2003 (the "Base Balance Sheet Date"), (x) set forth in Disclosure Schedule 3.05, (y) incurred under, or required or permitted to be incurred under, this Agreement, or (z) incurred in the ordinary course of business since the Base Balance Sheet Date, the Company has no material liabilities or obligations required to be disclosed or reserved against on a balance sheet prepared in accordance with generally accepted accounting principles.

-11-

3.06. ABSENCE OF CERTAIN CHANGES OR EVENTS. Except as set forth in Disclosure Schedule 3.06 or as permitted by Section 5.01 hereof, since the Base Balance Sheet Date and to and including the date of this Agreement:

(a) there has not been (i) a Material Adverse Effect, (ii) any declaration, setting aside or payment of any dividend or distribution (whether in cash, stock or property) in respect of the capital stock of the Company or any redemption or other acquisition by the Company of any of the capital stock of the Company or any split, combination or reclassification of shares of capital stock declared or made by the Company, (iii) any change in the amount or terms of wages, salary and other compensation paid to, or benefits provided to any current employee or current director of the Company (other than changes that occurred in the ordinary course of business or pursuant to plans, programs or agreements then existing), or termination of the employment of any employee of the Company, or (iv) any commitment or agreement to do any of the foregoing; and

(b) there have not been (i) any extraordinary losses suffered, (ii) any material assets mortgaged, pledged or made subject to any lien, charge or other encumbrance, (iii) any liability or obligation (absolute, accrued or contingent) incurred or any bad debt, contingency or other reserve increase suffered, except, in each such case, in the ordinary course of business and consistent with past practice, (iv) any claims, liabilities or obligations (absolute, accrued or contingent) paid, discharged or satisfied, other than the payment, discharge or satisfaction, in the ordinary course of business and consistent with past practice of claims, liabilities and obligations reflected or reserved against in the Financial Statements or incurred in the ordinary course of business and consistent with past practice since the Base Balance Sheet Date, (v) written off as uncollectible any notes or accounts receivable, except write-offs in the ordinary course of business and consistent with past practice, (vi) write down of the value of any asset or investment on the Company's books or records, except for depreciation and amortization taken in the ordinary course of business and consistent with past practice, (vii) any material

change in any method of accounting or accounting practice by the Company, except for such changes required by reason of changes in generally accepted accounting principles, (viii) cancellation of any debts or waiver of any claims or rights in excess of \$50,000, or sale, transfer or other disposition of any properties or assets (real, personal or mixed, tangible or intangible) in excess of \$50,000, except, in each such case, in transactions in the ordinary course of business and consistent with past practice, (ix) any single capital expenditure or commitment (other than expenditures under or related to software development contracts) in excess of \$50,000 for additions to property or equipment, or aggregate capital expenditures and commitments in excess of \$100,000 for additions to property or equipment, (x) any transaction entered into by the Company other than in the ordinary course of business, or (xi) any agreement to do any of the foregoing.

#### 3.07. TAX MATTERS.

(a) The Company has timely filed or caused to be filed all Federal, state, local and foreign Tax Returns required to be filed by it since January 1, 1998 with respect to Taxes and has paid all Taxes due and owing by the Company (whether or not shown on

-12-

any Tax Return) since such date and all assessments made against it to the extent such have become due. None of the foregoing Tax Returns contains any position which is or would be subject to penalties under Code Section 6662. In addition, the Company has paid or made adequate provision for the payment of all such Taxes which are due and payable with respect to the periods covered by such returns pursuant to any assessment with respect to such Taxes in such jurisdictions, whether or not in connection with such returns or otherwise due or payable on or before the Closing Date. The liability for Taxes reflected in the Base Balance Sheet (i) is sufficient for the payment of all unpaid Taxes, whether or not disputed, that are accrued or applicable for the period ended on the Base Balance Sheet Date and for all years and periods ended prior thereto and (ii) adjusted for the passage of time in a manner consistent with the past practice of the Company, is sufficient for the payment of all unpaid Taxes, whether or not disputed, that are accrued or applicable for all years or periods ending on or prior to the Closing Date.

- (b) All Tax Returns filed by the Company for any taxable period ending after January 1, 1998 were, as of the date filed, complete and accurate in all material respects. Except as set forth in Disclosure Schedule 3.07(b), no Tax Returns filed by the Company, the due date of which (taking into account all extensions of time to file) was on or after January 1, 1998, have been audited and no claims for additional Taxes for any taxable period have been made by any taxing authority that have not been resolved in full. The Company has not received a notice of deficiency or assessment of additional Taxes, which notice or assessment has not been resolved in full. The Company has not received from any taxing authority (including jurisdictions where the Company has not filed Tax Returns) any (i) notice indicating any attempt to open an audit or other review, or (ii) request for information relating to Tax matters. The Company has not waived any statute of limitations in respect of Taxes or extended the period for assessment or payment of any Tax, which period has not since expired.
- (c) The Company has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, former employee, independent contractor, creditor, Shareholder or other third party.
- (d) The Company has not been a member of an affiliated group (as such term is defined in Section 1504 of the Code) filing a consolidated federal income tax return for any tax year, or portion thereof, since the date of incorporation of the Company.
- (e) There are no liens for Taxes (other than current Taxes not yet due and payable) upon the assets of the Company.
- (f) The Company has not filed a consent under Code Section 341(f) concerning collapsible corporations.
- (g) The Company has not been a United States real property holding corporation within the meaning of Code Section 897(c)(2) during the applicable period specified in Code Section 897(c)(1)(A)(ii).

-13-

(h) The Company is not a party to (nor will the Company, prior to the Closing, become a party to) any Tax allocation, indemnity or sharing agreement.

- (i) There is no contract, agreement, plan or arrangement covering any employee or former employee of the Company that, individually or in the aggregate, could give rise to the payment (taking into consideration any payment made under Section 1.08 of this Agreement) of any amount that would not be deductible pursuant to Section 280G of the Code.
- (j) The Company has not entered into any transaction that is considered a "reportable transaction" under Treasury Regulation Section  $1.6011-4\,(\mathrm{b})$  .
- (k) The Company has not had any "ownership change" as that phrase is defined in Code Section 382.
- (1) The Company and the Shareholders have not taken (and will not take from the date of this Agreement to the Closing date) any action which would cause the Merger to fail to qualify for tax-free reorganization treatment under Code Section  $368\,(a)$ .
- (m) Disclosure Schedule 3.07(b) sets forth, as of the most recent practicable date, (as well as on an estimated pro forma basis as of the Closing giving effect to the consummation of the transactions contemplated hereby) the amount of any net operating loss, net capital loss, unused investment credit or other credit.
- (n) For purposes of this Agreement, the term "Tax" or "Taxes" means all taxes, including without limitation, any income, corporation, gross receipts, profits, gains, capital stock, capital duty, franchise, withholding, social security, unemployment, disability, property, wealth, welfare, stamp, excise, occupation, sales, use, value added, alternative minimum, estimated, or other similar tax (including any fee, assessment or other charge in the nature of or in lieu of any tax) imposed by any governmental entity (whether Federal, state, local, municipal, foreign or otherwise) or political subdivision thereof, and any interest, penalties, additions to tax or additional amounts in respect of the foregoing, and including any transferee or secondary liability in respect of any tax (whether imposed by law, contractual agreement or otherwise) and any liability in respect of any tax as a result of being a member of any affiliated, consolidated, combined, unitary or similar group. The term "Tax Return" means all returns, declarations, reports, statements and other documents required to be filed in respect of Taxes, and any claims for refunds of Taxes, including any amendments or supplements to any of the foregoing.
- 3.08. REAL PROPERTY. The real property demised by the leases described in Disclosure Schedule 3.08 (the "Leases") constitutes all of the real property owned, used or occupied by the Company (the "Real Property"). The Company does not own any Real Property. The Real Property has access, sufficient for the conduct of the Company's business as now conducted, to public roads and to all utilities, including electricity, sanitary and storm sewer, potable water, natural gas and other utilities, used in the operation of the business of the Company at that location. Each of the Leases is in full force and effect. Neither the Company

-14-

nor, to the knowledge of the Company, any other party thereto is in material default under or in respect of any of the Leases.

- 3.09. TITLE TO PROPERTIES. The Company owns or holds by valid lease or license all of its personal property reflected in the Base Balance Sheet or acquired after the Base Balance Sheet Date (except for any assets sold in accordance with Section 3.06(b) (viii), and assets sold after the Base Balance Sheet Date in the ordinary course of business and consistent with past practice), free and clear of all mortgages, claims, liens, security interests, charges and encumbrances, except (i) liens which may arise for current Taxes and assessments not yet due and payable or which are being contested in good faith and in respect of which adequate reserves have been established, (ii) imperfections in title and liens easements, and other liens, claims and encumbrances which were not incurred in connection with the borrowing of money or the obtaining of advances or credit and which do not materially detract from the value of or materially interfere with the present use of the properties subject thereto or affected thereby, and (iii) liens which are disclosed in Disclosure Schedule 3.09 ("Permitted Liens").
- 3.10. MATERIAL CONTRACTS. Except as set forth in Disclosure Schedule 3.10, the Company is not a party to or bound by any:
  - (a) contract with any labor union or any collective bargaining agreement,  $% \left( 1\right) =\left( 1\right) \left( 1\right)$
  - (b) bonus, pension, profit sharing, retirement, deferred compensation, savings, stock purchase, stock option, hospitalization, insurance or other plan providing similar employee benefits or

compensation;

- (c) employment, agency, consulting or similar service contract:
- (d) sales representative, distributorship, or similar agreement:
- (e) lease, whether as lessor or lessee, with respect to any real property;
- (f) contract as licensor or licensee for the license of any patent, know-how, trademark, trade name, service mark, copyright or other intangible asset (other than non-negotiated licenses of commercially available computer software);
- (g) loan or guaranty agreement, indenture or other instrument, contract or agreement under which any money has been borrowed or loaned or any note, bond or other evidence of indebtedness has been issued (other than trade accounts payable or receivable and other indebtedness incurred in the ordinary course of business and not for money borrowed);
- (h) mortgage, security agreement, sale-leaseback agreement or other agreement which effectively creates a lien on any assets of the Company;
- (i) contract restricting the Company in any material respect from engaging in business or from competing with any other parties;

-15-

- (j) purchase or sale order for merchandise or supplies which (i) was not entered into in the ordinary course of business, involves payments of \$100,000 or more, and is not terminable by the Company without cost or penalty upon sixty (60) days' or less notice, or (ii) is a standing or similar order with a remaining term of more than one (1) year, that involves payments of \$100,000 or more, and is not terminable by the Company without cost or penalty upon sixty (60) days' or less notice;
  - (k) plan of reorganization;
- (1) contract involving the acquisition or disposition of \$50,000 or more in assets, other than contracts involving the purchase or sale of inventory or products in the ordinary course of business;
- (m) partnership, limited liability company or joint venture agreement;
- (n) contract or commitment to loan money to any person, to guarantee indebtedness of any person, or to make an equity investment in any person;
  - (o) escrow agreement for software;
- (p) any other contract (excluding purchase and sale orders not required by the terms in clause (j) above to be set forth in Disclosure Schedule 3.10) that is not otherwise set forth in Disclosure Schedule 3.10, except such contracts as (x) involve payments of \$50,000 or less a year and (y) are terminable by the Company without cost or penalty upon sixty (60) days' or less notice; or
- $\mbox{\ensuremath{\mbox{\sc (q)}}}$  any contract requiring the consent of a third party upon a change in control.

All of the foregoing are hereinafter collectively called "Material Contracts." To the extent Material Contracts are evidenced by documents, true and correct copies thereof have been delivered or made available to the Parent unless otherwise noted in Disclosure Schedule 3.10. Except as set forth in Disclosure Schedule 3.10, (i) the Material Contracts are valid and enforceable in accordance with their respective terms with respect to the Company and, to the knowledge of the Company, are valid and enforceable in accordance with their respective terms with respect to any other party thereto, and (ii) there is not under any of the Material Contracts any existing breach, default or event of default by the Company or, to the knowledge of the Company, any other party thereto.

### 3.11. INTELLECTUAL PROPERTY

(a) As used in this Section 3.11, "Intellectual Property Rights" means patents, patent applications, inventions, trademarks (registered and unregistered), trademark applications, service marks (registered and unregistered), service mark applications, business names, design rights, copyrights (registered and unregistered), domain

names and Know-How. As used in this Section 3.11, "Know-How" means any and all technical or business information of the type generally treated as proprietary and confidential by a business, including, but not limited to, trade secrets, manufacturing processes, quality

-16-

control processes, drawings, product specifications, research and development results, customer lists and other customer information, supplier information, regulatory information, business methods and financial information. As used in this Section 3.11, "IP Embodiments" means all embodiments or uses of the Intellectual Property Rights, tangible or intangible (including, for example, but not limited to, each item of computer software and databases.

- (b) Disclosure Schedule 3.11 identifies each patent, registered trademark, trade name and registered copyright as well as all IP Embodiments that the Company owns or has the right to use pursuant to license, sublicense, agreement, or permission for operation of the businesses of the Company as presently conducted and as presently proposed to be conducted. The Company owns or has a valid right to use all Intellectual Property Rights that are used by the Company, all Intellectual Property Rights that are necessary to the operation of the business of the Company, and all Intellectual Property Rights that are embodied in the IP Embodiments of the Company. All Intellectual Property Rights and all IP Embodiments owned or used by the Company immediately prior to the Closing hereunder will be owned or available for use by the Parent on identical terms and conditions immediately subsequent to the Closing hereunder. The Company has taken all action necessary to maintain and protect all Intellectual Property Rights and all IP Embodiments related to the operation of the businesses of the Company that the Company owns or uses.
- (c) To the knowledge of the Company, the Company has not interfered with, infringed upon, misappropriated, or otherwise come into conflict with any Intellectual Property Rights of third parties related to the operation of the businesses of the Company. To the knowledge of the Company, neither the past nor current use of the IP Embodiments of the Company (i) has violated or infringed upon, or is violating or infringing upon, the rights of any person or other party; (ii) breaches any duty or obligation owed to any person or other party; or (iii) violates the privacy or any law relating to the privacy of any person or third party. The Company has not received any charge, complaint, claim, demand, or notice alleging any such interference, infringement, misappropriation, or violation (including any claim that the Company must license or refrain from using any Intellectual Property Right or IP Embodiment of any third party). To the knowledge of the Company, (a) no third party has interfered with, infringed upon, misappropriated, or otherwise come into conflict with any Intellectual Property Rights of the Company, (b) the Intellectual Property Rights of the Company are not invalid or unenforceable, and (c) there are no grounds that the Intellectual Property Rights of the Company are invalid or unenforceable.
- (d) With respect to all Intellectual Property Rights and all IP Embodiments of the Company: (i) the Company possesses all right, title, and interest in and to the item, free and clear of any security interest, license, or other restriction; (ii) the item is not subject to any outstanding injunction, judgment, order, decree, ruling, or charge; (iii) no action, suit, proceeding, hearing, investigation, charge, complaint, claim, or demand is pending or is threatened which challenges the legality, validity, enforceability, use, or ownership of the item; (iv) at no time has the Company ever agreed to indemnify any third party for or against any interference, infringement, misappropriation, or other

-17-

conflict with respect to the item; (v) there are no liens, licenses or other encumbrances affecting any Intellectual Property Rights or IP Embodiments other than those identified on Disclosure Schedule 3.11; and (vi) the Company will make available to the Parent correct and complete copies of all IP Embodiments at Closing.

(e) Disclosure Schedule 3.11(a) identifies all material Intellectual Property Rights and IP Embodiments that any third party owns and that the Company uses pursuant to license, sublicense, agreement, or permission. The Company has delivered to the Parent correct and complete copies of all such material licenses, sublicenses, agreements, and permissions (as amended to date). With respect to all Intellectual Property Rights and IP Embodiments identified in Disclosure Schedule 3.11(a): (i) the license, sublicense, agreement, or permission covering the item is legal, valid, binding, enforceable, and in full force and effect; (ii) the license, sublicense, agreement, or permission will continue to be legal, valid, binding, enforceable, and

in full force and effect on identical terms following the consummation of the transactions contemplated hereby; (iii) no party to the license, sublicense, agreement, or permission is in material breach or default, and no event has occurred which with notice or lapse of time would constitute a breach or default or permit termination, modification, or acceleration thereunder; (iv) no party to the license, sublicense, agreement, or permission has repudiated any material provision thereof; (v) with respect to each sublicense, the representations and warranties set forth in subsections (i) through (iv) above are true and correct with respect to the underlying license; (vi) there is no outstanding injunction, judgment, order, decree, ruling, or charge; and (vii) no action, suit, proceeding, hearing, investigation, charge, complaint, claim, or demand is pending or, to the knowledge of the Company, is threatened which challenges the legality, validity, or enforceability of any Intellectual Property Right.

- (f) To the knowledge of the Company, the Company will not interfere with, infringe upon, misappropriate, or otherwise come into conflict with any Intellectual Property Right of third parties as a result of the continued operation of its businesses as presently conducted and as presently proposed to be conducted.
- (g) Upon Parent's reasonable request, the Company and the employees of the Company will fully cooperate with Parent to file one or more U.S. provisional and non-provisional patent applications and associated PCT patent applications based on invention(s) conceived before Closing.
- (h) All IP Embodiments of the Company were, with respect to copyright, created as works made for hire (as defined under U.S. copyright law) or otherwise by employees of the Company. To the extent that any author or developer of any IP Embodiment of the Company was not an employee of the Company at the time such person contributed to such IP Embodiments, such author or developer has irrevocably assigned to the Company in writing all copyrights and other proprietary rights in such person's work with respect to such IP Embodiments.
- (i) With respect to the computer software listed on Disclosure Schedule 3.11: (i) the Company maintains machine-readable master-reproducible copies, source code

-18-

listings, technical documentation and user manuals for the most current releases or versions thereof and for all earlier releases or versions thereof currently being supported by them; (ii) in each case, the machine-readable copy substantially conforms to the corresponding source code listing; (iii) it is written in the language set forth on Disclosure Schedule 3.11, for use on the hardware set forth on Disclosure Schedule 3.11 with standard operating systems; (iv) it does not include any "open source" computer software, or any computer software that is subject to any "GNU Public License," or any other "free" software license containing equivalent provisions, (v) it can be maintained and modified by reasonably competent programmers familiar with such language, hardware and operating systems; (vi) in each case, it operates in accordance with all contractual requirements, specifications and documentation (including, but not limited to, any user documentation) without material operating defects, (vii) the Company has not received any communication that the computer software has failed, or is failing, to perform as represented or warranted to any licensee, (viii) the computer software shall, under normal use and service, record, store, process, present and communicate information and data with calendar dates falling on or after January 1, 2000 in the manner, and with the same functionality, as the computer records, stores, processes, presents and communicates information and data with calendar dates on or before December 31, 1999, (ix) except with respect to demonstration or trial copies, the computer software does not contain, and the Company has taken reasonable precautions to prevent the presence of, any malicious code, program, or other internal component (e.g., computer virus, computer worm, computer time bomb, or similar component) which could damage, destroy, or alter the computer software or other software, firmware, or hardware used by Parent or any customer, or which could, in any unintended manner, reveal, damage, destroy, or alter any data or other information accessed through or processed by the computer software.

(j) The Company has not disclosed or delivered to any escrow agent or to any other person or party, or permitted the disclosure to any escrow agent or to any other person or party of, the source code (or any aspect or portion thereof) for or relating to any past, present, or future product of any of the Company. No source code (or any aspect or portion thereof) has been provided to any third party, nor has any source code that Company has provided to an escrow agent been released from escrow to a third party or been the subject of a request for release from escrow.

(k) Any license, sublicense or other contract to which the Company is a party that covers or relates to any Intellectual Property Right or any IP Embodiment is, to the knowledge of the Company, legal, valid, binding, enforceable and in full force and effect with respect to the other party, and, upon consummation of the transactions contemplated hereby, will continue to be legal, valid, binding, enforceable and in full force and effect on terms identical to those in effect immediately prior to the consummation of the transactions contemplated hereby. No party is in material breach of or material default under any license, sublicense or other contract covering or relating to any Intellectual Property Right or IP Embodiment and has not performed any act or omitted to perform any act which, with notice or lapse of time or both, will become or result in a material violation, material breach or material default thereunder. No litigation is pending or to the knowledge of the Company threatened against the Company which challenges the

-19-

legality, validity, enforceability or ownership of any license, sublicense or other contract covering or relating to any Intellectual Property Right.

- (1) No Intellectual Property Right or IP Embodiment is owned by or registered in the name of any current or former owner, shareholder, partner, director, executive, officer, employee, salesman, agent, customer, representative or contractor of any of the Company, nor does any such person have any interest therein or right thereto, including the right to royalty payments.
- (m) There is no governmental prohibition or restriction on the use of any of the Intellectual Property Rights or IP Embodiments of the Company in any jurisdiction or on the export or import of any of the Intellectual Property Rights or IP Embodiments of the Company from or to any jurisdiction, in each case, as used, exported or imported by the Company.
- 3.12. LITIGATION. Except as set forth in Disclosure Schedule 3.12 no litigation, arbitration, or administrative proceeding or investigation is pending or, to the knowledge of the Company, threatened against the Company or any of its officers, employees or directors in connection with the business or affairs of the Company. Disclosure Schedule 3.12 lists each litigation, arbitration or administrative proceeding or investigation against the Company or any of its officers, employees or directors in connection with the business and affairs of the Company during the past five (5) years.
- 3.13. PERMITS, LICENSES, AUTHORIZATIONS; COMPLIANCE WITH LAWS. The Company has all material licenses, franchises, permits and other governmental authorizations necessary to conduct its business, and the Company is not in material violation of any such license, franchise, permit or other governmental authorization, or any material violation of any statute, law, ordinance, rule, regulation, judgment, order or decree applicable to it or any of its properties.

### 3.14. RETIREMENT AND BENEFIT PLANS.

- (a) Disclosure Schedule 3.14(a) lists each material employee pension, retirement, profit sharing, stock bonus, stock option, stock purchase, bonus, incentive, deferred compensation, hospitalization, medical, dental, vision, vacation, insurance, sick pay, disability, severance, or other employee benefit plans, funds, programs, policies, contracts or arrangements (hereinafter referred to as the "Company Benefit Plans"), including without limitation any Employee Benefit Plan (as hereinafter defined), that the Company maintains or to which the Company contributes or in which any current or former employee of the Company has accrued any benefits which they remain entitled to receive. Except as set forth in Disclosure Schedule 3.14(a):
  - (i) To the knowledge of the Company, each such Company Benefit Plan (and each related trust, insurance contract, or fund) complies in form and in operation in all material respects with the applicable requirements of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), the Code, and other applicable laws.

-20-

(ii) All contributions (including all employer contributions and employee salary reduction contributions) which are due have been paid to each such Employee Benefit Plan which is an Employee Pension Benefit Plan (as hereinafter defined) and all contributions for any period ending on or before the Closing Date which are not yet due will have been paid to each such Employee Pension Benefit Plan or accrued in accordance with the past custom and practice of the Company.

All material premiums or other payments for all periods ending on or before the time of Closing have been paid or accrued with respect to each such Employee Benefit Plan which is an Employee Welfare Benefit Plan (as hereinafter defined).

- (iii) Each Employee Pension Benefit Plan which is intended to be a "qualified plan" under Section 401(a) of the Code is so qualified.
- (iv) The Company does not currently maintain, nor has it ever maintained, an Employee Pension Benefit Plan which is subject to Title IV of ERISA, including without limitation any Multiemployer Plan (as hereinafter defined).
- (v) The Company has delivered or otherwise made available to Parent correct and complete copies of all plan documents and summary plan descriptions, the most recent determination letter received from the Internal Revenue Service, if any, the most recent Form 5500 Annual Report, if any, and all related trust agreements, insurance contracts and other funding agreements which implement each Employee Benefit Plan.
- (b) To the knowledge of the Company, there have been no Prohibited Transactions (as defined in ERISA Section 406 or Code Section 4975) for which the Company would be liable with respect to any Employee Benefit Plan. The Company has no liability for breach of fiduciary duty or any other failure to act or comply in connection with the administration or investment of the assets of any Employee Benefit Plan. No action suit, proceeding, hearing, examination, audit or investigation with respect to the administration or the investment of the assets of any Company Benefit Plan (other than routine claims for benefits) is pending or, to the knowledge of the Company, threatened.
- (c) The Company does not currently contribute to any Multiemployer Plan or have any liability (including withdrawal liability) under any Multiemployer Plan.
- (d) Except as set forth in Disclosure Schedule  $3.14\,(d)$ , the Company does not maintain and has not maintained or contributed to, or since such date been required to contribute to, any Employee Welfare Benefit Plan providing medical, health or life insurance or other welfare-type benefits for current or future retired or terminated employees, their spouses or their dependents (other than in accordance with Code Section 4980B).
  - (e) For purposes of this Section 3.14:

-21-

- (i) "Employee Benefit Plan" means any plan maintained by the Company for its employees which is (a) a nonqualified deferred compensation or retirement plan or arrangement which is an Employee Pension Benefit Plan, (b) a qualified defined contribution retirement plan or arrangement which is an Employee Pension Benefit Plan, (c) a qualified defined benefit retirement plan or arrangement which is an Employee Pension Benefit Plan (including any Multiemployer Plan), or (d) an Employee Welfare Benefit Plan or material fringe benefit plan or program.
- (ii) "Employee Pension Benefit Plan" has the meaning set forth in ERISA Sec. 3(2).
- (iii) "Employee Welfare Benefit Plan" has the meaning set forth in ERISA Sec.  $3\,(1)\,.$
- (iv) "Multiemployer Plan" has the meaning set forth in ERISA Sec. 3(37).
- (f) After the Effective Time, the Surviving Corporation will have no obligation to issue, transfer or sell any shares of capital stock of the Company, the Parent or the Surviving Corporation pursuant to any Company Benefit Plan, including the 2000 Plan.

### 3.15. EMPLOYEES.

(a) Disclosure Schedule 3.15(a) contains a list of all current employees of the Company as of the date of this Agreement (each a "Company Employee"), and each such Company Employee's (i) title and/or responsibility, (ii) compensation paid to such employee during the current fiscal year through November 30, 2003 and (iii) current annual compensation or hourly rates of pay, together with a statement of the full amount and nature of any other remuneration, whether in cash or in kind.

- (b) The employment relationship between the Company and each Company Employee is "employment at will."
- (c) As of the date of this Agreement, the Company is not the subject of any pending or, to the Company's knowledge, threatened proceeding alleging that the Company has engaged in any unfair labor practice under any state or federal law, statute or regulation. The Company is not a party to any collective bargaining agreement, nor, to Company's knowledge, is any labor union or other bargaining representative engaged in or seeking to be engaged in collective bargaining with respect to Company Employees. There is no pending or, to the Company's knowledge, threatened labor strike, dispute, walk-out, work stoppage, slow-down or lockout with respect to any Company Employees, and no such strike, dispute, walk-out, slow-down or lockout has occurred within the past five years.

-22-

- (d) As of the date of this Agreement, no group of Company Employees or any current Company Employee having total annual cash compensation of more than \$50,000\$ has or have given notice of intent to terminate employment with the Company.
- (e) The Company is not a party to or bound by any plan, agreement, arrangement or understanding, verbal or written, that concerns any term or condition of employment and relates to the employment of any Company Employee or any former employee of the Company.
- (f) Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (i) result in any material payment (including, without limitation, severance, unemployment compensation, golden parachute or otherwise) becoming due to any Company Employee or former employee of the Company, under any compensation and benefit plan, agreement, arrangement or understanding, whether verbal or written, (ii) materially increase the amount of benefits payable under any compensation and benefit plan, agreement, arrangement or understanding, whether verbal or written, or (iii) result in the acceleration of the time of payment or vesting of any such benefits to any material extent.
- (g) There are no workers' compensation claims pending against the Company, nor, to the Company's knowledge, does any circumstance exist that is reasonably likely to result in such a claim.
- (h) No current Company Employee is on a short or long-term disability leave, or other leave of absence.
- (i) To the knowledge of the Company, the Company has complied with all applicable local, state and federal laws and regulations relating to employment, including but not limited to laws relating to discrimination, hours of work, and the payment of wages or overtime wages, and there are no complaints, lawsuits or other proceedings pending against the Company brought by or on behalf of any applicant for employment, any Company Employee or former employee, or any class of the foregoing, relating to any such law or regulation, or alleging breach of any express or implied contract of employment, wrongful termination of employment, or alleging any other discriminatory, wrongful or tortious conduct in connection with the employment relationship.
- (j) There are no pending or, to the Company's knowledge, threatened investigations, audits, complaints, or proceedings against the Company by or before any federal, state, municipal or other governmental department, commission, board, bureau, agency, arbitration tribunal or instrumentality, whether domestic or foreign, respecting or involving any applicant for employment, any Company Employee or any former employee, or any class of the foregoing, including but not limited to:
  - (i) The Equal Employment Opportunity Commission or any other state or local agency with authority to investigate claims or charges of employment discrimination in the workplace;

-23

(ii) The United States Department of Labor or any other state or local agency with authority to investigate claims or charges in any way relating to the Family Medical Leave Act, the Sarbanes-Oxley Act, or hours of employment or wages:

Administration or any other state or local agency with authority to investigate claims or charges in any way relating to concerning the safety and health of Employees; and

- (iv) The Office of Federal Contract Compliance or any corresponding state agency.
- (k) Each employee hired after November 8, 1986, and employed in the United States, has completed and the Company has retained an Immigration and Naturalization Service Form I-9 in accordance with applicable rules and regulations. No Company Employee is (a) a non-immigrant employee whose status would terminate or otherwise be affected by the business transaction consummated under this Agreement, or (b) an alien who is authorized to work in the United States in non-immigrant status.
- (1) Each Company Employee has executed an "Employee Agreement Pertaining to Confidential Information, Improvements and Inventions, and Competitive Activities" substantially in the form attached hereto as Schedule 3.15(1) to this Agreement.

### 3.16. ENVIRONMENTAL MATTERS.

- (a) For purposes of this Agreement:
- (i) "Environmental Law" means any and all applicable foreign, federal, state and local laws, rules, requirements and regulations relating to pollution, the environment (including, without limitation, ambient air, surface water, groundwater, land surface or subsurface strata) or protection of human health as it relates to the environment including, without limitation, laws and regulations relating to releases of Hazardous Substances, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Substances or relating to management of asbestos; and
- (ii) "Hazardous Substance" means each of the following: (A) any petroleum or any by-products or fractions thereof, asbestos or asbestos-containing materials, urea formaldehyde foam insulation, any form of natural gas, explosives, polychlorinated biphenyls ("PCBs"), radioactive materials, ionizing radiation, electromagnetic field radiation or microwave transmissions; (B) any chemicals, materials or substances, whether waste materials, raw materials or finished products, which are now defined as or included in the definition of "hazardous substances," "hazardous wastes," "toxic substances," "toxic pollutants," "pollutants," "contaminants," or words of similar import under any Environmental Law; and (C) any other chemical, material or substance, whether

-24-

waste materials, raw materials or finished products, regulated or forming the basis of liability under any Environmental Law.

- (b) Except as set forth in Disclosure Schedule  $3.15\,(b)$  and the environmental assessments listed therein:
  - (i) The Company complies in all material respects with all Environmental Laws pertaining to the generation, storage, treatment and disposal of Hazardous Substances in the conduct of its business. No action, suit, proceeding, hearing, charge, complaint, claim or demand has been filed or commenced against the Company and is now pending alleging any such failure to comply;
  - (ii) The Company possesses (or has timely filed applications pending for) all material licenses and permits required by all Environmental Laws applicable to the ownership of its assets and the conduct of its business;
  - (iii) Since January 1, 1998, there has been no material spill, discharge, leak, emission, injection, disposal, escape, dumping or release of any Hazardous Substance by the Company on, under, above or into the environment in violation of Environmental Law;
  - (iv) No underground storage tanks are located on the Real Property which contain or, to the knowledge of the Company, previously contained any Hazardous Substances; and
  - (v) The Company has delivered to Parent true and complete copies of the environmental assessments listed on

- 3.17. INSURANCE. The Company is insured under the insurance programs listed in Disclosure Schedule 3.17. With respect to such insurance policies: (a) each policy is legal, valid, binding, enforceable and in full force and effect; (b) such policies will continue to be legal, valid, binding, enforceable and in full force and effect on identical terms following the consummation of the transactions contemplated hereby; (c) the Company is not in material breach or material default (including with respect to the payment of premiums or the giving of notices), and no event has occurred which, with notice or lapse of time, would constitute such a material breach or material default, or permit termination, modification or acceleration, under the policy; and (d) to the knowledge of the Company, no party to the policy has repudiated any provision thereof. The Company has been covered during the past five years by insurance in scope and amount customary and reasonable for the business in which it has engaged during such period.
- 3.18. CUSTOMER RELATIONSHIPS. The Company has not received written notice from any current customer that such current customer intends, and, to the knowledge of the Company, no such customer intends, to discontinue purchases of products and services in any material amount.

-2.5-

#### 3.19. PRODUCT WARRANTY.

- (a) The Company has not, within the last five (5) years, (i) incurred any material claims of any type in respect of or in connection with the ownership or use of its products or (ii) incurred any material expenses related to product warranty claims.
- (b) The Company has no material liabilities for replacement or repair of products sold, leased or delivered by the Company or for other damages in connection with product warranty claims. No product sold, leased or delivered by the Company is subject to any guaranty, warranty or other indemnity beyond the applicable standard terms and conditions of sale or lease. Disclosure Schedule 3.21(b) includes copies of the standard terms and conditions pursuant to which the Company sells generally its products.
- 3.20. CONDITIONS AFFECTING BUSINESS. There is no development or threatened development known to the Company, other than any such developments or threatened developments that (i) affect the Company's industry generally or (ii) relate to general economic conditions, with respect to the markets, products, services, clients, customers, facilities, computer software, databases, personnel, vendors, suppliers, operations, assets or financial condition of the Company which would have a Material Adverse Effect on the business and operations of the Company.
- 3.21. NO BROKERS OR FINDERS. The Company has not engaged any investment banker, broker or finder in connection with the transactions contemplated hereby.
- 3.22. DISCLOSURE. None of the representations or warranties contained in this Article III (as qualified by the Disclosure Schedules), taken as a whole, contains or will contain any untrue statement of a material fact, or omits any material fact the omission of which would make the statements made herein, in light of the circumstances in which they were made, misleading.
- 3.23. SOLE REPRESENTATIONS AND WARRANTIES. Neither the Company nor any of the Shareholders shall be deemed to have made to Parent or Merger Subsidiary any representation or warranty other than as expressly made in this Article III. In particular, without limiting the generality of this Section 3.23, neither the Company nor any of the Shareholders makes any representation or warranty with respect to (a) any projections, estimates or budgets previously delivered or made available to Parent or Merger Subsidiary concerning future revenues, expenses, expenditures or results of operations or (b) any other information or documents made available to Parent or Merger Subsidiary or their respective representatives with respect to the Company, except as expressly covered in this Article III.

-26

### ARTICLE IV

## REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUBSIDIARY

Parent and Merger Subsidiary represent and warrant to the Company as follows (for purposes of this Article IV, "the date of this Agreement" shall mean the date of the Original Merger Agreement):

4.01. ORGANIZATION, GOOD STANDING, EQUITY OWNERSHIP. Each of Parent and Merger Subsidiary is a corporation duly incorporated, validly existing and in good standing under the laws of its state of incorporation. Parent owns all of

the issued and outstanding capital stock of Merger Subsidiary, and Merger Subsidiary does not own capital stock of any corporation. Parent has delivered to the Company a copy of the respective articles or certificate of incorporation and bylaws of Parent and Merger Subsidiary. Each such copy is complete and correct as of the date hereof.

- 4.02. AUTHORIZATION AND EXECUTION. Each of Parent and Merger Subsidiary has the corporate power and authority to execute and deliver this Agreement and consummate the transactions contemplated hereby. The execution, delivery and performance of this Agreement by each of Parent and Merger Subsidiary have been duly and effectively authorized by the Boards of Directors of Parent and Merger Subsidiary and by Parent as the sole shareholder of Merger Subsidiary, and no further corporate action is necessary on the part of Parent or Merger Subsidiary to consummate the transactions contemplated hereby. This Agreement constitutes a legal, valid and binding obligation of each of Parent and Merger Subsidiary, enforceable against each of them in accordance with its terms, except to the extent that enforceability may be limited by applicable bankruptcy, insolvency or similar laws affecting the enforcement of creditors' rights generally, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a court of law or equity).
- 4.03. NO CONFLICTS. Neither the execution and delivery of this Agreement by Parent and Merger Subsidiary, nor the consummation of the transactions contemplated hereby will (i) conflict with or result in a breach of the articles or certificate of incorporation or bylaws, as currently in effect, of Parent or Merger Subsidiary, or, (ii) except for the filing of the Articles of Merger as required by the MBCA and any filing required pursuant to the HSR Act, require any filing with, or the consent or approval of, any governmental authority having jurisdiction over any of the business or assets of Parent or Merger Subsidiary, or (iii) violate any statute, regulation, injunction, judgment or order to which Parent or Merger Subsidiary is subject, or (iv) result in a breach of, or constitute a default or an event which, with the passage of time or the giving of notice, or both, would constitute a default, give rise to a right of termination, cancellation or acceleration, create any entitlement to any payment or benefit, require the consent of any third party or result in the creation of any lien on the assets of Parent or Merger Subsidiary under any material instrument, contract or agreement to which either of Parent or Merger Subsidiary is a party or by which either of them is bound.

-27-

4.04. LITIGATION. No litigation, arbitration or administrative proceeding is pending or, to the knowledge of Parent or Merger Subsidiary, threatened against Parent or Merger Subsidiary as of the date of this Agreement that seeks to enjoin or otherwise challenges the consummation of the transactions contemplated by this Agreement.

### 4.05. SEC FILINGS; FINANCIAL STATEMENTS.

- (a) Parent has filed, and made available to the Company and its representatives, all forms, reports and documents required to be filed by it with the SEC since January 1, 2001 through the date of this Agreement (collectively, the "Parent SEC Reports"). As of the respective dates they were filed, (i) the Parent SEC Reports were prepared, and all forms, reports and documents filed with the SEC after the date of this Agreement and prior to the Effective Time will be prepared, in all material respects in accordance with the requirements of the Securities Act or the Securities Exchange Act of 1934, as amended (the "Exchange Act") as the case may be, and (ii) none of the Parent SEC Reports contained, nor will any forms, reports and documents filed after the date of this Agreement and prior to the Effective Time contain, any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading, except to the extent superseded by a Parent SEC Report filed subsequently and prior to the date hereof. Parent is eligible to register the resale of the Parent Company Stock to be issued hereby to certain Shareholders under a Form S-3 registration statement.
- (b) Each of the consolidated financial statements (including, in each case, any notes thereto) contained in the Parent SEC Reports and in any form, report or document filed after the date of this Agreement and prior to the Effective Time was, or will be, as the case may be, prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated (except as may be indicated in the notes thereto or, in the case of unaudited statements, as permitted by GAAP) and each presented or will present fairly, in all material respects, the consolidated financial position of Parent and its consolidated subsidiaries as at the respective dates thereof and for the respective periods indicated therein, except as otherwise noted therein (subject, in the case of unaudited statements, to normal year-end adjustments that would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect (as hereinafter defined)).

- 4.06. NO BROKERS OR FINDERS. Neither Parent nor Merger Subsidiary has engaged any investment banker, broker or finder in connection with the transactions contemplated hereby.
- 4.07. INTERIM OPERATIONS OF MERGER SUBSIDIARY. Merger Subsidiary was formed on December 22, 2003, solely for the purpose of engaging in the transactions contemplated hereby, has engaged in no other business activities and has conducted its operations only as contemplated hereby.

-28-

4.08. SOLE REPRESENTATIONS AND WARRANTIES. Neither Parent nor Merger Subsidiary nor any of their respective stockholders shall be deemed to have made to the Company or the Shareholders any representation or warranty other than as expressly made in this Article IV.

#### ARTICLE V

#### CONDUCT AND TRANSACTIONS PRIOR TO THE EFFECTIVE TIME

### 5.01. OPERATION OF BUSINESS OF THE COMPANY UNTIL EFFECTIVE TIME.

The Company covenants and agrees that, from the date hereof to the Effective Time, the Company will, except as required in connection with the Merger and the other transactions contemplated by this Agreement, and except as otherwise set forth in, or expressly permitted or contemplated by, this Agreement, disclosed in the Exhibits or Disclosure Schedules hereto, or consented to in writing by Parent:

- (a) Carry on its businesses in the ordinary and regular course, consistent with past practice and in substantially the same manner previously conducted, and use its commercially reasonable good faith efforts to preserve intact its business organization and the goodwill of its customers, supplier and others having business relations with the Company;
  - (b) Not amend its Articles of Incorporation or Bylaws;
- (c) Not issue, sell or grant options, warrants or rights to purchase or subscribe to, or enter into any arrangement or contract with respect to the issuance or sale of, any of the capital stock of the Company or rights or obligations convertible into or exchangeable for any shares of the capital stock of the Company or make any changes (by split-up, combination, reorganization or otherwise) in the capital structure of the Company;
- (d) Not declare, pay or set aside for payment any dividend or other distribution in respect of the capital stock of the Company and not redeem, purchase or otherwise acquire any shares of the capital stock or other securities of the Company or rights or obligations convertible into or exchangeable for any shares of the capital stock or other securities of the Company or obligations convertible into such, or any options, warrants or other rights to purchase or subscribe to any of the foregoing, except to the extent specifically contemplated by Sections 1.06 and 5.19 of this Agreement;
- (e) Not acquire or enter into an agreement to acquire, by merger consolidation or purchase of stock or assets, any business or entity:
- (f) Not (i) create, incur or assume any long-term debt (including obligations in respect of capital leases which individually involve annual payments in excess of \$50,000 or \$100,000 in the aggregate) or create, incur or assume any short-term debt for borrowed money (except in the ordinary course of business under existing lines of credit), (ii) assume, guarantee, endorse or otherwise become liable or responsible (whether

-29-

directly, contingently or otherwise) for the obligations of any other person, (iii) make any loans or advances to any other person, or (iv) make any capital contributions to, or investments in, any person;

(g) Not enter into, modify or extend in any manner the terms of any employment, severance or similar agreements with any Company Employee or current director of the Company, nor make any change in the amount or terms of wages, salary and other compensation paid to, or benefits provided to any Company Employee or current director of the Company (other than changes that occur in the ordinary course of business or pursuant to plans, programs or agreements then existing), including any such change pursuant to any stock option, incentive or bonus, stock purchase, pension, profit-sharing, deferred compensation, retirement or other plan, arrangement, contract, understanding or

- (h) Not permit a change in its methods of maintaining its books, accounts or business records or, except as required by generally accepted accounting principles (in which event prior notice shall be given to Parent), change any of its accounting principles, periods or the methods by which such principles are applied for Tax or financial reporting purposes;
- (i) Not make or change any election with respect to Taxes or consent to any waiver or extension of time to assess or collect any Taxes; and
- (j) Prepare and file all federal, state, local and foreign Tax Returns required to be filed by it, on a timely basis and in a manner consistent with past practice; and pay all Taxes when due, and withhold all Taxes when and as required, under all applicable law. Parent shall have a reasonable opportunity to review all such Tax Returns prior to filing.
- 5.02. SHAREHOLDER APPROVAL. The Company shall either (i) cause a special meeting of the Shareholders to be duly called and held, or (ii) send to each Shareholder a Written Consent in Lieu of Meeting, in either case as soon as reasonably practicable after the execution of this Agreement for the purpose of voting on the approval of this Agreement. Approval by the Shareholders in either manner set forth above by the vote required by the MBCA and the Company's Articles of Incorporation and Bylaws is referred to herein as the "Shareholder Approval."
- 5.03. NO SHOPPING. From the date hereof until the earliest to occur of (i) the Effective Time, (ii) the date of termination of this Agreement pursuant to Section 8.01 hereof and (iii) February 28, 2004, the Company will not, and will use reasonable efforts not to, permit any officer, director or other agent or representative of the Company to, directly or indirectly, (i) take any action to seek, initiate or solicit any offer from any person or group to acquire any shares of capital stock of the Company, to merge or consolidate with the Company, or to otherwise acquire, except to the extent not prohibited by Section 5.01 hereof, any significant portion of the assets of the Company, taken as a whole (a "Third Party Acquisition Offer"), or (ii) engage in negotiations concerning a Third Party Acquisition Offer with any person or group, or disclose financial information relating to the Company or any confidential or proprietary trade or business information relating to the business of the Company, or afford access to the

-30-

properties, books or records of the Company, to any person or group that the Company has reason to believe may be considering a Third Party Acquisition Offer.

- 5.04. ACCESS TO INFORMATION. From the date hereof until the Effective Time, the Company will give Parent and its counsel, financial advisors, auditors and other authorized representatives reasonable access, during normal business hours throughout the period prior to the Effective Time, to the offices, properties, books and records of the Company all at reasonable times and upon reasonable notice, and will instruct the employees, counsel, financial advisors and auditors of the Company to cooperate with Parent and each such representative in all reasonable respects in its investigation of the business of the Company. Parent and each such representative will conduct such investigation in a manner as not to interfere unreasonably with the operations of the Company and will take all necessary precautions (including obtaining the written agreement of its respective employees or representatives involved in such investigation) to protect the confidentiality of any information of the Company disclosed to such persons during such investigation. Prior to the Effective Time, Parent and its representatives shall not contact or communicate with the customers and suppliers of the Company in connection with the transactions contemplated by this Agreement except with the prior written consent of the Company, which consent shall not unreasonably be withheld.
- 5.05. CONFIDENTIALITY AGREEMENT. The Confidentiality Agreement between the Company and Parent dated June 18, 2003 shall remain in full force and effect until the Effective Time. Until the Effective Time, the Company and Parent agree to comply with the terms of such Confidentiality Agreement.
- 5.06. DEPOSIT OF ESCROWED FUNDS. At the Effective Time, the Escrow Agreement shall be executed and delivered by the parties thereto, and Parent shall deposit with the Escrow Agent the Escrowed Funds by delivering cash by wire transfer of same day funds. The Escrowed Funds shall remain deposited with the Escrow Agent and shall be subject to and payable in accordance with the terms of the Escrow Agreement and this Agreement.
- 5.07. REASONABLE EFFORTS. Subject to the terms and conditions of this Agreement, the parties hereto shall each use their reasonable best efforts to perform their obligations herein and to take, or cause to be taken or do, or cause to be done, all things necessary, proper or advisable under applicable law

to obtain all regulatory approvals and satisfy all conditions to the obligations of the parties under this Agreement and to cause the Merger and the other transactions contemplated herein to be carried out promptly in accordance with the terms hereof and shall cooperate fully with each other and their respective officers, directors, employees, agents, counsel, accountants and other designees in connection with any steps required to be taken as part of their respective obligations under this Agreement.

5.08. AMENDMENT OF DISCLOSURE SCHEDULES. From time to time prior to the Effective Time, the Company may supplement or amend the Disclosure Schedules in order to make the information set forth therein timely, complete and accurate. For purposes of determining the fulfillment of the condition set forth in Section 6.01(a) as of the Closing, the Disclosure Schedules shall be deemed to include only that information contained therein as of the date of the Original Merger Agreement and shall be deemed to exclude any information contained in any subsequent supplement or amendment thereto. For purposes of determining the

-31-

accuracy of the representations and warranties contained in Article III and the liability of the Company with respect thereto under Article X, the Disclosure Schedules shall be deemed to include all information contained in any supplement or amendment thereto made on or before the Closing.

5.09. EMPLOYMENT AGREEMENTS. The Company shall use commercially reasonable efforts to cause up to six senior executives of the Company mutually acceptable to Parent and the Company (collectively, the "Key Employees") to enter into employment-related agreements in a form and according to terms and conditions satisfactory to Parent, the material economic terms of which are described in Exhibit B hereto. Each such agreement shall become effective at the Closing and shall provide for the termination or amendment, effective immediately prior to the Closing, of any other agreement between the Company and the Key Employees affecting their employment by the Company.

#### 5.10. INCENTIVE PROGRAMS.

- (a) Prior to the Effective Time, Parent and the Company shall have adopted and approved an incentive program substantially in the form attached hereto as Exhibit C ("Incentive Program 1"), effective as of the Effective Time, for eligible Company Employees as described in Incentive Program 1.
- (b) Prior to the Effective Time, Parent and the Company shall have adopted and approved an incentive program substantially in the form attached hereto as Exhibit D (the "Incentive Program 2"), effective as of the Effective Time, for eligible Company Employees as described in Incentive Program 2.
- 5.11. RETENTION PROGRAM. Prior to the Effective Time, Parent and the Company shall have adopted and approved a retention program substantially in form attached hereto as Exhibit E (the "Retention Program"), effective as of the Effective Time, for eligible Company Employees.
- 5.12. REGISTRATION AND LISTING OF SHARES. Shares of Parent Common Stock delivered to the Shareholders shall be listed with the NYSE and registered pursuant to a Form S-3 registration statement (the "Form S-3") to be declared effective by the SEC. Parent shall file the Form S-3 within five (5) business days following the Closing Date. To the extent the Form S-3 is not reviewed by the Securities and Exchange Commission (the "SEC"), Parent shall file, no later than three (3) business days following the SEC's determination not to review the Form S-3, an acceleration request with the SEC requesting that effectiveness of the Form S-3 occur as soon as practicable, but in any event no later than the expiration of such three-day period. If the SEC determines to review the Form S-3, then Parent shall promptly respond to any comments received from the SEC, and shall use its commercially reasonable efforts to clear SEC comments and obtain effectiveness of the Form S-3 no later than ninety (90) days following the date upon which the Form S-3 initially was filed with the SEC.
- 5.13. RESTRICTIVE LEGEND ON SHARES. Upon the Effective Time, each certificate representing Shares of Parent Company Stock shall be endorsed with the following legend:

-32-

"THE SHARES REPRESENTED BY THIS CERTIFICATE HAVE BEEN ACQUIRED FOR INVESTMENT AND NOT WITH A VIEW TO, OR IN CONNECTION WITH, THE SALE OR DISTRIBUTION THEREOF. NO SUCH SALE OR DISPOSITION MAY BE EFFECTED WITHOUT AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933."

Parent agrees that it will take all appropriate actions necessary to remove the legend promptly after the registration statement referred to in Section 5.12 is declared effective.

- 5.14. TAXES ON TRANSFER. The Shareholders shall be responsible for any taxes resulting from any transfer of the Shares of Parent Company Stock to the Shareholders, including any transfer, documentary, sales, use, stamp, registration or other similar taxes and fees (including any interest due).
- 5.15. CONDITIONS TO PARENT'S OBLIGATION TO FILE REGISTRATION STATEMENT. Parent's obligation to file a registration statement shall, in all cases, be subject to the following provisions:
  - (a) The Shareholders shall have agreed to furnish any and all agreements, consents and representations required of them by the SEC in connection with the filing or which are a condition of having the subject registration statement declared effective.
  - (b) Parent shall not be obligated to file any registration statement if Parent (i) shall have delivered to the Shareholder proposing to sell shares of Parent Common Stock delivered pursuant to this Agreement an opinion of counsel, which counsel shall be acceptable to such Shareholder, to the effect that the shares of Parent Common Stock delivered pursuant to this Agreement may lawfully be sold to the public without registration under the Securities Act and (ii) has delivered to its transfer agent instructions to register the transfer of any of the shares of Parent Common Stock sold in reliance on such opinion.
  - (c) Parent shall have received all information reasonably requested in writing by Parent from the Shareholders that is necessary for the filing of the registration statement.
- 5.16. INVESTMENT REPRESENTATIONS. The Company shall use its commercially reasonable efforts to cause each Shareholder receiving Parent Shares to execute a representation letter to Parent prior to the Effective Time in which such Stockholder represents that such Stockholder is an "accredited investor" as defined in Regulation D under the Securities Act of 1933, as amended, which representation letter shall be in a form reasonably acceptable to Parent.
- 5.17. STOCK OWNERSHIP PLAN. Following the Closing, the Company and the trustees of the Company's Stock Ownership Plan shall take, or cause to be taken, all action necessary to promptly terminate the Stock Ownership Plan and dissolve the Stock Ownership Plan trust; provided, however, that such termination and dissolution shall not occur prior to the

-33-

registration of the Shares of Parent Company Stock issued to the Stock Ownership Plan pursuant to the Merger and the removal of all restrictions on the sale or disposition of such Shares.

- 5.18. NON-SOLICITATION AGREEMENTS. The Company shall use commercially reasonable efforts to cause the holders of shares of the Company's issued and outstanding preferred stock to enter into one or more agreements with the Surviving Corporation and/or Parent, in a form and according to terms and conditions reasonably satisfactory to Parent, each of which shall become effective at the Closing and which shall require the holders of shares of the Company's issued and outstanding preferred stock, for a period of time mutually acceptable to Parent and such preferred shareholders, to not solicit, hire or enter into any arrangement for the services of any Company Employee who remains an employee of the Surviving Corporation, New LLC or any affiliate thereof immediately following the Effective Time, subject to customary exceptions and qualifications.
- 5.19. REDEMPTION OF SHARES. Prior to the Effective Time, the Company shall use its best efforts to cause the redemption for cash of all shares of the Company's Common Stock held by each Shareholder who holds less than 2,000 shares of the Company's Common Stock as of the date of the Original Merger Agreement; provided, however, that additional Shareholders owning shares of the Company's Common Stock may be asked by the Company to redeem their shares prior to the Effective Time solely for cash if reasonably required in order to comply with the federal securities laws. Such redemption shall be accomplished in compliance with all applicable securities laws. Parent shall fund the amount of cash redemption paid by the Company pursuant to this Section 5.19 as part of the aggregate merger consideration; provided, however, that (a) in no event shall the aggregate merger consideration paid by Parent exceed \$68,000,000, and the updated versions of Schedules 1.05, 1.06(a) and Schedule 1.08 to be agreed upon by the parties as contemplated by Article I shall reflect the transactions and payments contemplated by this Section 5.19, and (b) in no event shall the dollar thresholds set forth in Section 1.06 be decreased as a result of the cash paid by the Company pursuant to this Section 5.19.
- 5.20. RESIGNATION OF EMPLOYEES. Prior to the Effective Time, the Company shall secure a written agreement, to take effect within 60 days after the Effective Time, by each of Jeffrey L. Laurel and Gregory A. Twedt, to resign his employment with the Company, with no further liability or obligations by the Company or New LLC to either Laurel or Twedt, except as set forth under this

#### ARTICLE VI

### CONDITIONS PRECEDENT

- 6.01. CONDITIONS TO THE OBLIGATIONS OF PARENT AND MERGER SUBSIDIARY. The obligations of Parent and Merger Subsidiary to effect the Merger shall be subject to the fulfillment at or prior to the Effective Time of the following conditions, any one or more of which may be waived by Parent and Merger Subsidiary:
  - (a) The representations and warranties of the Company contained in Article III of this Agreement shall be true and correct as of the date of the Original  ${\sf N}$

-34-

Merger Agreement and there shall be no inaccuracy in any such representation or warranty immediately prior to the Effective Time (except with respect to representations and warranties that refer to or speak as of a specified date, which shall only need to have been true on and as of such date), except to the extent any such inaccuracy would not impair the ability of the Company to consummate the transactions contemplated hereby or have, in the aggregate, a Material Adverse Effect; the Company shall have performed and complied in all material respects with the agreements and obligations contained in this Agreement required to be performed and complied with by it immediately prior to the Effective Time; and Parent and Merger Subsidiary shall have received a certificate signed by the Chief Executive Officer of the Company to the effects set forth in this Section 6.01(a).

- (b) The Shareholder Approval shall have been obtained, as set forth in Section 5.02, and shall remain in full force and effect.
- (c) There shall not be pending any litigation or administrative proceeding brought by any governmental or other regulatory or administrative agency or commission requesting or looking toward an injunction, writ, order, judgment or decree which, in the reasonable judgment of Parent, is reasonably likely, if issued, to restrain or prohibit the consummation of any of the transactions contemplated hereby or require rescission of this Agreement or any such transactions or result in material damages to Parent, Merger Subsidiary or the Surviving Corporation or their respective officers or directors if the transactions contemplated hereby are consummated, nor shall there be in effect any injunction, writ, judgment, preliminary restraining order or other order or decree of any nature issued by a court or governmental agency of competent jurisdiction directing that any of the transactions provided for herein not be consummated as so provided.
- (d) All other corporate action on the part of the Company necessary to authorize the execution, delivery and consummation of this Agreement or any agreement or instrument contemplated hereby to which the Company is or is to be a party or the transactions contemplated hereby or thereby shall have been duly and validly taken.
- (e) From the date hereof until the Effective Time, there shall not have been any damage, destruction, loss or casualty to property or assets of the Company which has had, or could reasonably be expected to have, a Material Adverse Effect.
- (f) The consents, authorizations, orders and approvals listed on Schedule 6.01(f), including approval under the HSR Act, shall have been obtained or made.
- (g) The Escrow Agreement shall have been executed and delivered by the Company and the Escrow Agent.
- (h) Parent shall have received the resignation of all directors of the Company or such directors shall have otherwise been removed in accordance with applicable law.
- (i) Since the date of the Original Merger Agreement, there shall not have been a material adverse effect upon the business, operations, properties or financial condition of Parent (a "Parent Material Adverse Effect"); provided, however, that a

-35-

Parent Material Adverse Effect will not be deemed to have occurred if the change, circumstance, event, effect or state of facts results primarily from (a) a change in general economic conditions, (b) a change in general business conditions in Parent's industry that does not disproportionately affect Parent or (c) the disclosure to the

public, or pendency, of the Merger and the transactions contemplated thereby.

- (j) In the reasonable, good faith judgment of Parent, the offering and issuance of shares of Parent Common Stock to certain Shareholders, as contemplated under Article I of this Agreement, shall be exempt from registration under the federal securities laws in reliance upon Rule 506 of Regulation D under the Securities Act of 1933, as amended.
- 6.02. CONDITIONS TO THE OBLIGATIONS OF THE COMPANY. The obligations of the Company to effect the Merger shall be subject to the fulfillment at or prior to the Effective Time of the following conditions, any one or more of which may be waived by the Company:
  - (a) The representations and warranties of Parent and Merger Subsidiary contained in Article IV of this Agreement shall be true and correct in all material respects as of the date of the Original Merger Agreement and immediately prior to the Effective Time as if such representations and warranties had been made on and as of the Effective Time; each of Parent and Merger Subsidiary shall have performed and complied in all material respects with the agreements and obligations contained in this Agreement required to be performed and complied with by it immediately prior to the Effective Time; and the Company shall have received a certificate signed by a duly authorized executive officer of each of Parent and Merger Subsidiary to the effects set forth in this Section 6.02(a).
  - (b) The Shareholder Approval shall have been obtained as set forth in Section 5.02, and shall remain in full force and effect.
  - (c) There shall not be pending any litigation or administrative proceeding brought by any governmental or other regulatory or administrative agency or commission requesting or looking toward an injunction, writ, order, judgment or decree which, in the reasonable judgment of the Company, is reasonably likely, if issued, to restrain or prohibit the consummation of any of the transactions contemplated hereby or require rescission of this Agreement or any such transactions or result in material damages to the Company, its officers or directors, or the Shareholders if the transactions contemplated hereby are consummated, nor shall there be in effect any injunction, writ, judgment, preliminary restraining order or other order or decree of any nature issued by a court or governmental agency of competent jurisdiction directing that any of the transactions provided for herein not be consummated as so provided.
  - (d) All corporate action on the part of Parent and Merger Subsidiary necessary to authorize the execution, delivery and consummation of this Agreement or any agreement or instrument contemplated hereby to which Parent or Merger Subsidiary is or is to be a party or the transactions contemplated hereby or thereby shall have been duly and validly taken.

-36-

- (e) All consents, authorizations, orders and approvals of (or filings or registrations with) any governmental commission, board or other regulatory body (domestic or foreign) required in connection with the execution, delivery and performance of this Agreement, including approval under the HSR Act, shall have been obtained or made, except for filing of the Articles of Merger and any other documents required to be filed after the Effective Time, except where the failure to have obtained or made any such consent, authorization, order, approval, filing or registration would not have a Material Adverse Effect. All consents required from third parties in order for the Company to consummate the Merger shall have been obtained except for such consents, the failure of which to obtain, would not have a Material Adverse Effect.
- (f) The Escrow Agreement shall have been executed and delivered by Parent and the Escrow Agent.
- $\mbox{\em (g)}$  There shall not have been a Parent Material Adverse Effect.

# ARTICLE VII

## CONDUCT AND TRANSACTIONS AFTER THE EFFECTIVE TIME

7.01. INDEMNIFICATION. All rights to indemnification, expense advancement and exculpation existing in favor of any present or former director, officer or employee of the Company as provided in the Articles of Incorporation, Bylaws or similar organizational documents of the Company or Bylaw as in effect on the date hereof shall survive the Merger for a period of at least six (6) years after the Effective Time (or, in the event any relevant claim is asserted

or made within such six-year period, until final disposition of such claim) with respect to matters occurring at or prior to the Effective Time, and no action taken during such period shall be deemed to diminish the obligations set forth in this Section 7.01. Parent hereby guarantees, effective at the Effective Time, all obligations of the Surviving Corporation or New LLC in respect of such indemnification and expense advancement.

- 7.02. DIRECTORS AND OFFICERS LIABILITY INSURANCE. For a period of at least six (6) years after the Effective Time, Parent shall cause the Surviving Corporation and New LLC to maintain in effect a run-off (i.e., "tail") policy or endorsement with respect to the current policy of directors' and officers' liability insurance covering claims asserted within six (6) years after the Effective Time arising from facts or events which occurred at or before the Effective Time (including consummation of the transactions contemplated by this Agreement); and such policies or endorsements shall name as insureds thereunder all present and former directors and officers of the Company; provided that such coverage and limits are readily available in the insurance market and that the premium for such policy shall not exceed three times the current premium.
- 7.03. CONFIDENTIALITY. After the Closing, Shareholders shall strictly maintain the confidentiality of all proprietary information of the Company ("Confidential Information"), except to the extent disclosure of any such information is reasonably believed by the disclosing party to be required by law or authorized by the other party (which, for the Shareholders, shall be the Shareholder Representative), or otherwise made publicly available by the other party, or reasonably occurs in connection with disputes over the terms of this Agreement. In the event

-37-

that Shareholders reasonably believe after consultation with counsel that they are required by law to disclose any Confidential Information described in this Section 7.03, the Shareholder Representative will (a) use its commercially reasonable efforts to provide the Parent with reasonably prompt notice before such disclosure in order that the Parent may attempt to obtain a protective order or other assurance that confidential treatment will be accorded to Confidential Information, and (b) cooperate with the Parent in attempting to obtain such order or assurance. The provisions of this Section 7.03 shall not apply to any information which is in the public domain or shall come into the public domain, other than by reason of fault by the Shareholders or becomes known in the industry through no wrongful act on the part of the Shareholders. In addition, the parties hereto agree and acknowledge that nothing in this Section 7.03 shall prohibit a Shareholder from disclosing to its officers, directors, partners, managers or shareholders the specific terms and conditions of the transactions to be consummated pursuant to this Agreement.

- 7.04. EMPLOYMENT FOLLOWING EFFECTIVE TIME. All Company Employees employed immediately prior to the Effective Time shall remain employed by the Surviving Corporation, New LLC or an affiliate thereof immediately following the Effective Time (a "Surviving Employee"). The parties hereto acknowledge that the employment of any Surviving Employee may be transferred to Parent, or to an affiliate of any of Parent, the Surviving Corporation or New LLC. Nothing in this Agreement, however, is intended to or shall be interpreted to require Parent, the Surviving Corporation, New LLC or an affiliate of any of Parent, the Surviving Corporation or New LLC to continue the employment of any Surviving Employee for any period of time following the Effective Time.
- 7.05. COMPLETION OF MERGER BETWEEN NEW LLC AND THE COMPANY. As soon as possible, but in any event no later than the close of business on the first business day immediately following the Effective Time, the Company, as part of a single plan of reorganization of which the Merger is a part, shall be merged with and into New LLC in accordance with the provisions of Minnesota and Delaware law. New LLC shall ultimately be the surviving entity in the Merger.

## ARTICLE VIII

# TERMINATION AND ABANDONMENT

- 8.01. GENERALLY. This Agreement may be terminated and abandoned at any time prior to the Effective Time:
  - (a) by mutual written consent of Parent and the Company (in the case of the Company, as approved by its Board of Directors);
  - (b) by Parent or the Company (in the case of the Company, as approved by its Board of Directors) if the transactions contemplated hereby shall not have been consummated on or before February 28, 2004 (which date may be extended by mutual agreement of Parent and the Company (in the case of the Company, as approved by its Board of Directors)), provided that such failure is not due to the failure of the party seeking to terminate this Agreement (or, in the event Parent is seeking to terminate this Agreement, of Merger Subsidiary) to comply with its obligations under this Agreement;

- (c) by Parent, if (i) any of the conditions set forth in Section 6.01 hereof shall become impossible to fulfill other than for reasons within the control of Parent or Merger Subsidiary, and such conditions shall not have been waived pursuant to Section 11.03 hereof; or (ii) the Shareholder Approval shall not have been obtained within 45 days after the signing of this Agreement; or
- (d) by the Company (as approved by its Board of Directors), if (i) any of the conditions set forth in Section 6.02 hereof shall become impossible to fulfill other than for reasons within the control of the Company, and such conditions shall not have been waived pursuant to Section 11.03 hereof; or (ii) the Shareholder Approval shall not have been obtained within 45 days after the signing of this Agreement.
- 8.02. PROCEDURE AND EFFECT OF TERMINATION AND ABANDONMENT. Upon termination of this Agreement by the Company or Parent pursuant to Section 8.01 hereof, written notice thereof shall immediately be given to the other such party and this Agreement shall terminate and the Merger shall be abandoned without further action by any of the parties hereto. If this Agreement is terminated as provided herein, this Agreement shall forthwith become void and no party hereto shall have any liability or further obligation to any other party to this Agreement, except to the extent the termination is a direct result of a willful and material breach or violation by such party of a representation, warranty or covenant contained in this Agreement and except for continuing obligations under Section 5.05.

#### ARTICLE IX

#### SHAREHOLDER REPRESENTATIVE

- 9.01. DESIGNATION. Subject to the terms and conditions of this Article IX, Michael Gorman and Jeffrey L. Laurel, and each of them acting alone, with full power of substitution and re-substitution, is designated by the Company, on behalf of each Shareholder, as the representative (either or both of such persons acting individually, when acting in such capacity, being defined herein as the "Shareholder Representative") of the Shareholders to serve, and Parent hereby acknowledges that the Shareholder Representative shall serve, as the sole representative of the Shareholders, from and after the Effective Time with respect to the matters set forth in this Agreement and the Escrow Agreement, such service to be without compensation except for the reimbursement of out-of-pocket expenses and indemnification specifically provided herein. The Shareholder Representative has accepted such designation as of the date hereof. Notwithstanding anything to the contrary contained in this Agreement or the Escrow Agreement, the Shareholder Representative shall have no duties or responsibilities except those expressly set forth herein, and no implied covenants, functions, responsibilities, duties, obligations or liabilities on behalf of any Shareholder shall otherwise exist against the Shareholder Representative.
- 9.02. AUTHORITY. Each of the Shareholders, by (a) voting in favor of this Agreement and the transactions contemplated herein at the special meeting of the Shareholders to be duly called and held, (b) approving this Agreement and the transactions contemplated herein by signing a Written Consent in Lieu of Meeting, or (c) accepting any portion of the consideration to be paid pursuant to Section 1.06 will, effective as of the Effective Time,

-39-

irrevocably appoint the Shareholder Representative as the agent, proxy and attorney-in-fact for such Shareholder for all purposes of this Agreement, including full power and authority on such member's behalf (i) to take all actions which the Shareholder Representative considers necessary or desirable in connection with the defense, pursuit or settlement of any determinations relating to the payment of the Cash Flow Holdback Indemnification Escrow Amount and any claims for indemnification pursuant to Article X hereof, including to sue, defend, negotiate, settle and compromise any such claims for indemnification made by or against, and other disputes with, Parent pursuant to this Agreement or any of the agreements or transactions contemplated hereby, (ii) to engage and employ agents and representatives (including accountants, legal counsel and other professionals) and to incur such other expenses as he shall deem necessary or prudent in connection with the administration of the foregoing, (iii) to provide for all expenses incurred in connection with the administration of the foregoing to be paid by directing the Escrow Agent to reimburse the Shareholder Representative for such expenses, (iv) to disburse to the Shareholders all indemnification payments received from Parent under Article X hereof, (v) to accept and receive notices to the Shareholders pursuant to this Agreement, and (vi) to take all other actions and exercise all other rights which the Shareholder Representative (in his sole discretion) considers necessary or appropriate in connection with this Agreement. Each of the Shareholders, by the adoption of this Agreement either at the special meeting of the Shareholders or by Written Consent in Lieu of Meeting referred to in Section 5.02, or by accepting any portion of the consideration to be paid pursuant to Section 1.06, agrees that such agency and proxy are coupled with an interest, and are therefore irrevocable without the consent of the Shareholder Representative and shall survive the death, incapacity, bankruptcy, dissolution

or liquidation of any Shareholder. All decisions and acts by the Shareholder Representative shall be binding upon all of the Shareholders, and no Shareholder shall have the right to object, dissent, protest or otherwise contest the same.

- 9.03. RESIGNATION. In the event that both individuals authorized hereunder as a Shareholder Representative shall die, become incapacitated, resign or otherwise fail to act on behalf of the Shareholders for any reason, the Shareholder Representative shall be such other person as shall be selected by Christopher D. Heim, and such substituted representative shall be deemed to be the Shareholder Representative for all purposes of this Agreement.
- 9.04. RELIANCE BY THIRD PARTIES ON THE SHAREHOLDER REPRESENTATIVE'S AUTHORITY. The Shareholder Representative is authorized to act on behalf of the Shareholders notwithstanding any dispute or disagreement among the Shareholders, and the other parties hereto shall be entitled to rely on any and all action taken by the Shareholder Representative without any liability to, or obligation to inquire of, any Shareholder even if such party shall be aware of any actual or potential dispute or disagreement among the Shareholders. Each of the other parties hereto is expressly authorized to rely on the genuineness of the signature of the Shareholder Representative and, upon receipt of any writing which reasonably appears to have been signed by the Shareholder Representative, the other parties hereto may act upon the same without any further duty of inquiry as to the genuineness of the writing.
- 9.05. EXCULPATION AND INDEMNIFICATION. Neither the Shareholder Representative nor any agent employed by him shall be liable to any Shareholder relating to the performance of his duties under this Agreement for any errors in judgment, negligence, oversight, breach of duty or otherwise except to the extent it is finally determined in a court of

-40-

competent jurisdiction by clear and convincing evidence that the actions taken or not taken by the Shareholder Representative constituted fraud or were taken or not taken in bad faith. The Shareholder Representative shall be indemnified and held harmless by the Shareholders against all Losses (as hereinafter defined) paid or incurred in connection with any action, suit, proceeding or claim to which the Shareholder Representative is made a party by reason of the fact that he was acting as the Shareholder Representative pursuant to this Agreement; provided, however, that the Shareholder Representative shall not be entitled to indemnification hereunder to the extent it is finally determined in a court of competent jurisdiction by clear and convincing evidence that the actions taken or not taken by the Shareholder Representative constituted fraud or were taken or not taken in bad faith; and provided further, however, that the Shareholder Representative shall have recourse only against the unpaid Indemnification Escrow Amount (fully subordinated in right of payment and otherwise to Parent's claims thereto, whether or not then existing or known), with respect to such Losses as provided in the next two sentences of this Section 9.05. Any amount owing to the Shareholder Representative from the Shareholders pursuant to this Section 9.05 shall be reduced on a pro rata basis from the next succeeding distribution(s) of the Indemnification Escrow Amount by the Escrow Agent to the Shareholders, and shall be payable solely from such source. The Shareholder Representative shall be protected in acting upon any notice, statement or certificate believed by him to be genuine and to have been furnished by the appropriate person and in acting or refusing to act in good faith or any matter.

### ARTICLE X

### INDEMNIFICATION

- 10.01. INDEMNIFICATION BY THE SHAREHOLDERS. The Shareholders and each of their successors and assigns (solely through the Indemnification Escrow Amount as set forth in Section 10.08(c)) shall indemnify, defend, reimburse and hold harmless Parent, Merger Subsidiary, New LLC, the Company (following the Effective Time), and each of their respective directors, officers, employees and agents (each a "Buyer Indemnified Party" and collectively, "Buyer Indemnified Parties") from and against any and all Losses (as defined in Section 10.10) resulting from or relating to (a) any breach of any representation or warranty of the Company contained in Article III of this Agreement or (b) the breach or failure to perform by the Company of any of its covenants or agreements set forth in Articles V and VII of this Agreement.
- 10.02. INDEMNIFICATION BY PARENT, MERGER SUBSIDIARY AND NEW LLC. Parent, Merger Subsidiary and New LLC shall indemnify, defend, reimburse and hold harmless the Shareholders, their successors and assigns and each of their respective directors, officers, employees and agents (each a "Shareholder Indemnified Party" and collectively, "Shareholder Indemnified Parties") from and against any and all Losses resulting from or relating to (a) any breach of any representation or warranty of Parent or Merger Subsidiary contained in this Agreement or (b) the breach or failure to perform by Parent or Merger Subsidiary of any of Parent's or Merger Subsidiary's covenants or agreements set forth in Article V and VII of this Agreement.

- 10.03. NOTICE OF THIRD-PARTY CLAIMS. In no case shall any Indemnitor (as defined in Section 10.10) under this Agreement be liable with respect to any Third-Party Claim (as defined Section 10.10) against any Indemnitee (as defined in Section 10.10) unless the Indemnitee shall have delivered to the Indemnitor a Claim Notice (as hereinafter defined) and the following conditions are satisfied:
  - (a) TIMELY DELIVERY OF CLAIM NOTICE. Except as provided in paragraphs (b) and (c) of this Section 10.03, and subject to Section 10.07 below, no right to indemnification under this Article X shall be available to the Indemnitee with respect to a Third-Party Claim unless the Indemnitee shall have delivered to the Indemnitor, with a copy to the Escrow Agent, within the Notice Period (as defined in Section 10.10) a notice ("Claim Notice") describing in reasonable detail the facts giving rise to such Third-Party Claim and stating that the Indemnitee intends to seek indemnification for such Third-Party Claim from the Indemnitor pursuant to this Article X.
  - (b) LATE DELIVERY OF CLAIM NOTICE. If, in the case of a Third-Party Claim, a Claim Notice is not given to the Indemnitor within the Notice Period, the Indemnitee shall nevertheless be entitled to be indemnified under this Article X (i) if the Indemnitee can establish that the time elapsed between the end of the Notice Period and the giving of the Claim Notice is reasonable in all the circumstances; or (ii) to the extent (but only to the extent) that the Indemnitee can establish that the Indemnitor has not been prejudiced by such time elapsed.
  - (c) PAID OR SETTLED CLAIMS. If a Claim Notice is not given by the Indemnitee prior to the payment or settlement by the Indemnitee of a Third-Party Claim, the Indemnitee shall only be entitled to be indemnified under this Article X to the extent (and only to the extent) that the Indemnitee can establish that the Indemnitor has not been prejudiced by such payment or settlement.
- 10.04. DEFENSE OF THIRD-PARTY CLAIMS. Upon receipt of a Claim Notice from an Indemnitee with respect to any Third-Party Claim, the Indemnitor may assume the defense thereof with counsel reasonably satisfactory to such Indemnitee and the Indemnitee shall cooperate in all reasonable respects in such defense. The Indemnitee shall have the right to employ separate counsel in any action or claim and to participate in the defense thereof, provided that the fees and expenses of counsel employed by the Indemnitee shall be at the expense of the Indemnitor only if such counsel is retained pursuant to the following sentence or if the employment of such counsel has been specifically authorized by the Indemnitor. If the Indemnitor does not notify the Indemnitee within thirty (30) days after receipt of the Claim Notice that the Indemnitor elects to undertake the defense thereof, the Indemnitee shall have the right to defend at the expense of the Indemnitor the claim with counsel of its choosing reasonably satisfactory to the Indemnitor, subject to the right of the Indemnitor to assume the defense of any claim at any time prior to settlement or final determination thereof. In such event, the Indemnitee shall send a written notice to the Indemnitor of any proposed settlement of any claim, which settlement the Indemnitor may reject, in its reasonable judgment within thirty (30) days of receipt of such notice. Failure to reject such notice within such thirty (30)-day period shall be deemed an acceptance of such notice. If the Indemnitor has assumed the defense of a claim from the Indemnitee, the Indemnitee shall have the right to settle any such claim over

-42-

the objection of the Indemnitor only if the Indemnitee waives any right to indemnity therefor. So long as the Indemnitor is conducting the defense of any Third-Party Claim in accordance with the terms hereof, the Indemnitee agrees that Indemnitor shall have full and complete control over the conduct of such proceeding, including any settlement thereof.

- 10.05. NOTICE OF OTHER CLAIMS. In the event any Indemnitee should have a claim against any Indemnitor hereunder that does not involve a Third-Party Claim being asserted against or sought to be collected from the Indemnitee, the Indemnitee shall notify the Indemnitor, with a copy to the Escrow Agent, with reasonable promptness of such claim by the Indemnitee, specifying the nature of and specific basis for such claim and the amount or the estimated amount of such claim. The Indemnitor shall remit (except if the Indemnitor is the Shareholder Representative acting on behalf of the Shareholders, in which case the Indemnitor shall instruct the Escrow Agent to remit) payment for the amount of such claim upon receipt of an invoice therefor, or in the event of a dispute, the Indemnitee and the Indemnitor shall proceed in good faith to negotiate a resolution of such dispute, and if not resolved through negotiations, such dispute shall be resolved in accordance with Section 11.16 hereof.
- 10.06. ACCESS AND COOPERATION. After the Closing Date, Parent, the Surviving Corporation and New LLC on the one hand, and the Shareholders, on the other hand, shall each cooperate fully with the other as to all claims for indemnification hereunder, shall make available to the other as reasonably

requested all information, books, records and documents that may be potentially relevant to any and all claims and shall preserve all such information, books, records and documents until the termination of any claim. Parent, the Surviving Corporation and New LLC, on the one hand, and the Shareholders, on the other hand, shall each also make available to the other, as reasonably requested, its personnel, agents and other representatives who are responsible for preparing or maintaining information, books, records or other documents, or who may have particular knowledge with respect to any claim.

10.07. TERM OF INDEMNITIES. Except as hereinafter expressly provided, the right to indemnification under Sections 10.01 and 10.02 hereof shall expire twelve (12) months following the Closing Date; provided, however, that such limitation shall not be applicable to an indemnification claim for which a Claim Notice was given within the twelve-month period following the Closing Date with respect to claims under Section 10.01, to the extent any such claim is pending upon expiration of the applicable period. Any such pending claim shall survive in respect of that claim until the final determination or settlement of that claim.

### 10.08. LIMITATIONS ON LIABILITY.

(a) REMEDIES EXCLUSIVE. Except as specifically provided for herein, each of the Parent, Merger Subsidiary, New LLC and the Shareholders acknowledge and agree that, should the Closing occur, the sole and exclusive remedy of each party hereto with respect to any and all claims arising out of this Agreement shall be pursuant to the indemnification provisions set forth in this Article X; provided, however, that this Section 10.08 shall not apply to Losses resulting from fraud or intentional misrepresentation.

-43-

### (b) BASKET.

- (i) No claim for indemnification under Sections 10.01 or 10.02 shall be made by Parent, a Buyer Indemnified Party or a Shareholder Indemnified Party with respect to any breach resulting in an individual item of Loss, or related items of Losses arising out of substantially similar facts and circumstances, unless and until the amount of such Losses suffered by the Buyer Indemnified Parties or the Shareholder Indemnified Parties exceeds \$15,000, at which point a claim can be made for the entire amount of such Losses, subject to the limitations set forth in Section 10.08(b)(ii); and
- (ii) Except as hereinafter expressly provided, no claim for indemnification under Sections 10.01 or 10.02 shall be made by a Buyer Indemnified Party or Shareholder Indemnified Party, as the case may be, unless and until the aggregate amount of all Losses suffered by the Buyer Indemnified Parties or the Shareholder Indemnified Parties exceeds \$500,000 (the "Basket"), and then the aggregate amount of all such Losses including the full amount of the Basket shall be recoverable by the Buyer Indemnified Party or Shareholder Indemnified Party, as the case may be; provided, however, that claims for indemnification arising solely from any breach of any representation or warranty of the Company set forth in Section 3.07 ("Tax Claims") may be made by a Buyer Indemnified Party at such time as the aggregate amount of all Losses suffered by the Buyer Indemnified Parties arising from Tax Claims exceeds \$100,000 (the "Tax Basket"), and then the aggregate amount of all such Losses including the full amount of the Tax Basket shall be recoverable by the Buyer Indemnified Party.
- (c) LIABILITY LIMITED TO INDEMNIFICATION ESCROW AMOUNT. Parent and the Surviving Corporation shall have recourse only against the undisbursed Indemnification Escrow Amount with respect to any Losses, in accordance with the terms of the Escrow Agreement, except in the event of fraud or intentional misrepresentation. Neither the Shareholders nor any of their successors and assigns shall have any liability separate from, or in addition to, the Indemnification Escrow Amount.
- (d) EFFECT OF ADJUSTMENTS AND RESERVES. Notwithstanding any other provision in this Agreement to the contrary, neither the Shareholders nor their successors and assigns shall have any liability for, and a Buyer Indemnified Party shall not bring any indemnification claim against any Shareholder or its successors or assigns for, any of the following: (i) a Loss, to the extent that the amount thereof was (A) included in the calculation of the merger consideration payable pursuant to Sections 1.05 and 1.06 or (B) the basis of an adjustment made to the merger consideration pursuant to Sections 1.05 and 1.06; or (ii) a Loss, to the extent that it is reflected or reserved for on the Actual Balance Sheet and incorporated into the calculation of an

-44-

(e) OTHER LIMITATIONS. Notwithstanding any of the foregoing,
(i) Parent, Merger Subsidiary and the Surviving Corporation shall have no recourse against the Company's officers, directors or agents other than in their capacities as Shareholders otherwise entitled to receive distributions from the Indemnification Escrow Amount under the Escrow Agreement, and (ii) indemnification owing from any Shareholder who is or was an officer or director of the Company shall not be deemed for any purpose to be a claim covered by indemnification owing to such Shareholder by the Company under any law, bylaw or agreement whatsoever

### 10.09. INDEMNIFICATION ESCROW AMOUNT AND TERM.

- (a) Except as provided in Section 10.09(b) below, the Escrow Agreement will provide that, on the date that is twelve (12) months from the Closing Date (the "Release Date"), the Escrow Agent shall release from escrow and deliver to the Shareholders the remaining Indemnification Escrow Amount and the Escrow Agreement shall terminate on such date.
- (b) If there are outstanding indemnification claims on the Release Date, and the remaining Indemnification Escrow Amount would not cover the amount of such pending claims, then an amount adequate to cover such pending claims shall be withheld by the Escrow Agent from the Indemnification Escrow Amount, and the Escrow Agreement shall continue, for as long as required to cover such pending claims.

### 10.10. INDEMNITY DEFINITIONS.

"Indemnitee" means any person who may be entitled to seek indemnification pursuant to the provisions of Sections 10.01 or 10.02 hereof (or the Shareholder Representative in the case of indemnification owing to the Shareholders).

"Indemnitor" means any person who may be obligated to provide indemnification pursuant to Sections 10.01 or 10.02 hereof (or the Shareholder Representative in the case of indemnification to be provided by the Shareholders).

"Loss" or "Losses" means any and all demands, claims, payments, obligations, actions or causes of action, assessments, losses, liabilities, damages (but, except with respect to any Third-Party Claim and/or a claim relating to fraud or intentional misrepresenation, excluding incidental, special, consequential, exemplary, punitive and similar damages), costs and expenses paid or incurred, including without limitation any legal or other expenses reasonably incurred in connection with investigating or defending any claims or actions and all amounts paid in settlement of claims or actions in accordance with Section 10.04 hereof. The amount of any Losses for which indemnification is provided under this Article X shall be net of any amounts recovered or recoverable by the Indemnitee under insurance policies with respect thereto (and thus any such amounts shall not be included in calculating the deductible set forth in Section 10.08 (b)) and shall be (i) increased to take account of any net tax costs actually incurred by the Indemnitee in connection with the receipt of indemnity payments hereunder and (ii) reduced to take into account the then present value of any net tax benefit (including as a result of any basis adjustment) realized or to be realized by the

-45-

Indemnitee in connection with any such Loss. In computing the amount of any such net tax cost or net tax benefit, the Indemnitee shall be deemed to recognize all other items of income, gain, loss, deduction or credit before recognizing any item arising from the receipt of any indemnity payment hereunder or the incurrence or payment of any Loss.

"Notice Period" as applied to any Third-Party Claim for which an Indemnitee seeks to be indemnified pursuant to Article X hereof, means the period ending on the earlier of the following:

(i) Six months after the time at which the chairman, president or any vice president or secretary (each a "Specified Officer") of the Indemnitee (or the Indemnitee, if the Indemnitee is an individual) has either (x) received actual notice of the facts giving rise to such Third-Party Claim or (y) commenced an active investigation of circumstances likely to give rise to such Third-Party Claim and, in each case, where such Specified Officer (or Indemnitee, if an individual) believes or should reasonably believe that such facts or circumstances could reasonably be expected to give rise to such

Third-Party Claim for which such Indemnitee would be entitled to Indemnification pursuant to Article X hereof; provided, however, in no event shall the Notice Period extend past the one year anniversary of the Closing Date;

- (ii) With respect to any Third-Party Claim that has become the subject of proceedings before any court or tribunal, such time as would allow the Indemnitor sufficient time to contest, on the assumption that there is an arguable defense to such Third-Party Claim, such proceeding prior to any judgment or decision thereon; or
- (iii) With respect to any Third-Party Claim that has become the subject of settlement proceedings, such time as would provide the Indemnitor sufficient time prior to such settlement to determine whether to contest such claim and assume the defense pursuant to Section 10.04 hereof.

"Third-Party Claims" means any and all Losses which arise out of or result from (i) any claims or actions asserted against an Indemnitee by a third party, (ii) any rights of a third party asserted against an Indemnitee, or (iii) any liabilities of, or amounts payable by an Indemnitee to a third party arising in respect of, claims, actions and rights referred to in the foregoing clauses (i) or (ii).

-46-

## ARTICLE XI

# MISCELLANEOUS PROVISIONS

- 11.01. SURVIVAL OF REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE COMPANY, PARENT AND MERGER SUBSIDIARY.
  - (a) The representations, warranties and covenants of the Company set forth in Articles III and V of this Agreement and in any directly related closing certificate or Disclosure Schedule furnished in connection herewith shall survive the Closing Date but shall expire on the date which is twelve (12) months following the Closing Date.
  - (b) The representations, warranties and covenants of Parent and Merger Subsidiary set forth in Articles IV and V of this Agreement and in any directly related closing certificate furnished in connection herewith shall survive until the date which is twelve (12) months following the Closing Date.
- 11.02. AMENDMENT AND MODIFICATION. To the extent permitted by applicable law, this Agreement may be amended, modified or supplemented only by written agreement of the parties hereto (in the case of the Company, as approved by its Board of Directors) at any time prior to the Effective Time with respect to any of the terms contained herein, except that after the Shareholder Approval contemplated by Section 5.02 hereof, the amount of the merger consideration shall in no event be decreased and the form of the merger consideration shall in no event be altered without the approval of the Shareholders.
- 11.03. WAIVER OF COMPLIANCE; CONSENTS. Any failure of Parent or Merger Subsidiary, on the one hand, or the Company, on the other hand, to comply with any obligation, covenant, agreement or condition herein (except the conditions in Sections 6.01(b) and 6.02(b) of this Agreement) may be waived in writing by the Company, or by Parent and Merger Subsidiary, respectively, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure. Whenever this Agreement requires or permits consent by or on behalf of any party hereto, such consent shall be given in writing in a manner consistent with the requirements for a waiver of compliance as set forth in this Section 11.03.
- 11.04. EXPENSES. All expenses incurred in connection with this Agreement and the consummation of the transactions contemplated hereby shall be paid by the party incurring such expenses; provided, however, that the legal and accounting fees of the Company shall be payable in accordance with the provisions set forth in Section 1.06 of this Agreement.

11.05. PRESS RELEASES AND PUBLIC ANNOUNCEMENTS. No party to this Agreement shall issue any press release or make any announcement or other verbal or written communication to employees, third parties or the public relating to the subject matter of this Agreement without prior written approval of the other parties; provided, however, that each of the Company and Parent may make any public disclosure it reasonably believes in good faith (based upon written opinion of counsel) is required by applicable law or, in the case of Parent,

-47-

any listing or trading agreement concerning its publicly traded securities (in which case the disclosing party will advise the other parties to this Agreement prior to making the disclosure).

11.06. ADDITIONAL AGREEMENTS. In case at any time after the Effective Time any further action is necessary or desirable to carry out the purposes of this Agreement, the proper officers and directors of each corporation which is a party to this Agreement shall take all such necessary action.

11.07. NOTICES. All notices and other communications hereunder shall be in writing and shall be deemed given (a) if delivered personally, effective when delivered, or (b) if sent by facsimile, telecopy or other electronic transmission device, when receipt is acknowledged; provided, however, that if receipt is acknowledged after normal business hours of the recipient, notice shall be deemed to have been given on the next business day, or (c) if delivered by express delivery service, effective when delivered, or (d) if mailed by registered or certified mail (return receipt requested), effective three business days after mailing, to the parties at the following addresses (or at such other address for a party as shall be specified by like notice):

If to Parent or Merger Subsidiary, to it at or c/o:

3M Company Building 220-4W-03 3M Center St. Paul, MN 55144 Facsimile: (651) 736-5516

Attention: Division Vice President
Industrial Services and Solutions Division

with a copy to:

3M Company Building 220-14W-07 3M Center St. Paul, MN 55144 Facsimile: (651) 736-7859

Attention: Senior Vice President Legal Affairs

-48-

If to the Company, to it at:

HighJump Software, Inc. 6455 City West Parkway Eden Prairie, MN 55344 Facsimile: (952) 947-0440

Attention: Chief Executive Officer

with a copy to:

Dorsey & Whitney LLP 50 South Sixth Street Suite 1500 Minneapolis, MN 55402 Facsimile: (612) 340-7800

Attention: Kenneth L. Cutler

11.08. ASSIGNMENT. This Agreement and all of the provisions hereof shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other parties. Except for the provisions of Article I and Sections 7.01 and 7.02 hereof, this

Agreement is not intended to confer upon any other person except the parties hereto any rights or remedies hereunder.

11.09. INTERPRETATION. As used in this Agreement, unless otherwise expressly defined herein, (i) the term "including" shall mean "including without limitation"; (ii) the term "person" shall mean and include an individual, a partnership, a limited liability company, a joint venture, a corporation, a trust, an incorporated organization and a government or any department or agency thereof; (iii) the term "affiliate" shall have the meaning set forth in Rule 12b-2 of the General Rules and Regulations promulgated under the Securities Exchange Act of 1934, as amended; (iv) the phrase "business day" shall mean any day other than a Saturday, Sunday or a day which is a statutory holiday under the laws of the United States or the State of Illinois; (v) all dollar amounts are expressed in United States funds; and (vi) the phrase "to the knowledge of the Company" or any similar phrase shall mean the knowledge of one or more of the following officers of the Company: Chris Heim (the Company's President and Chief Executive Officer), and Dan Mayleben (the Company's Chief Financial Officer), including facts of which a reasonably prudent person in a similar position should have known ,following due inquiry.

### 11.10. DISPUTE RESOLUTION.

(a) Any disagreement or dispute (a "Dispute") between the parties arising out of or related to this Agreement shall be resolved in the manner provided in this Section 11.10. In the event of a Dispute, either party may, by written notice to the other party, request that such Dispute be referred to the Division Vice President of Industrial Services

-49-

and Solutions Division of Parent and the Shareholder Representative, who shall negotiate in good faith to attempt to resolve the Dispute. No settlement reached under this Section 11.10 shall be binding on the parties until reduced to a writing signed by the parties.

- (b) If the procedure outlined in Section 11.10 fails to bring about a resolution of each outstanding Dispute within 30 days following notice thereof, then the parties shall promptly initiate a voluntary, non-binding mediation conducted by a mutually agreed upon mediator in Minneapolis, Minnesota. If the parties do not agree upon a mediator, they shall request the American Arbitration Association to appoint a capable mediator for them in accordance with the Commercial Dispute Resolution Procedures of the American Arbitration Association in effect as of the date of this Agreement. Parent and Shareholder Representative shall each bear one-half of the costs and expenses of the mediation and shall endeavor in good faith to resolve therein each outstanding Dispute. No settlement reached under this Section 11.10 shall be binding upon the parties until reduced to a writing signed by the parties. In the event the parties are unable to resolve any outstanding Dispute as provided above, then such outstanding Dispute shall be determined by litigation conducted in accordance with this Agreement.
- 11.11. GOVERNING LAW. This Agreement shall be governed by the laws of the State of Minnesota without giving effect to conflict or choice of laws principles that may be applicable.
- 11.12. COUNTERPARTS. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 11.13. HEADINGS; INTERNAL REFERENCES. The Article and Section headings contained in this Agreement are solely for the purpose of reference, and are not part of the agreement of the parties and shall not affect in any way the meaning or interpretation of this Agreement.
- 11.14. NUMBER; GENDER. Whenever the context so requires, the singular shall include the plural and the plural shall include the singular, and the gender of any pronoun shall include the other genders.
- 11.15. ENTIRE AGREEMENT. This Agreement, including the Exhibits and the Disclosure Schedules hereto and the Confidentiality Agreement described in Section 5.05 hereof, embody the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein, and supersede all prior agreements and understandings among the parties with respect to such subject matter, including but not limited to the Original Merger Agreement. There are no restrictions, promises, representations, warranties (express or implied), covenants or undertakings of the parties, other than those expressly set forth or referred to in this Agreement or such Confidentiality Agreement.

# 11.16. ENFORCEMENT.

(a) The parties hereto agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with its terms or were

otherwise breached. Each party shall be entitled to injunctive relief to prevent any breach of this Agreement and to enforce this Agreement specifically in any court of the State of Minnesota or any court of the United States located in the State of Minnesota (in each case in addition to any other remedy to which such party is entitled at law or in equity).

- (b) In addition, each party hereby:
- (i) submits itself to the personal jurisdiction of (A) the courts of the State of Minnesota; and (B) the United States District Court for the District of Minnesota with respect to any dispute arising out of this Agreement or any transaction contemplated hereby to the extent such courts would have subject matter jurisdiction with respect to such dispute;
- (ii) agrees that it will not attempt to deny or defeat such personal jurisdiction or venue by motion or other request for leave from any such court; and
- (iii) agrees that it will not bring any action relating to this Agreement (or any transactions contemplated by this Agreement) in any court other than the courts referred to in Section 11.16(b)(i) above.
- 11.17. WAIVER. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO TRIAL BY JURY IN CONNECTION WITH ANY LITIGATION ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE ESCROW AGREEMENT OR THE MERGER OR ANY OF THE OTHER TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.
- 11.18. SEVERABILITY. If any term, provision, covenant, agreement or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants, agreements and restrictions of this Agreement will continue in full force and effect and will in no way be affected, impaired or invalidated.

## 11.19. DISCLOSURE SCHEDULES.

- (a) Matters reflected in the Disclosure Schedules are not necessarily limited to matters required by this Agreement to be reflected in the Disclosure Schedules. Such additional matters are set forth for informational purposes and do not necessarily include other matters of a similar nature.
- (b) A disclosure made by the Company in any Section of this Agreement or any Disclosure Schedule that is sufficient to reasonably inform the Parent of information required to be disclosed in another Section of this Agreement or another Disclosure Schedule in order to avoid a misrepresentation thereunder shall be deemed, for all purposes of this Agreement, to have been made with respect to such other Section of this Agreement or such Disclosure Schedule. In no event shall the mere listing in a Disclosure Schedule (or the mere provision by the Company to the Parent of a copy) of a document or other item be deemed adequate to disclose an exception to a representation or warranty made herein (unless the representation or warranty has to do with the

-51-

existence of the document or other item itself, or unless the listing of a document or item in a Disclosure Schedule otherwise reasonably informs the Parent of an exception to such representation or warranty).

# 11.20. DEFINITIONS.

Term	Section or Place Where Defined
2000 Plan Additional Closing Cash Consideration affiliate Aggregate Cash Amount	1.06(b) 11.09 1.05(c)
Aggregate Consideration	Preamble
Articles of Merger	3.05
Basket. business day.	10.08(b)
Buyer Indemnified Parties	10.01
Cash Flow Holdback	1.09(b)

# SIGNATURE PAGE TO AMENDED AGREEMENT AND PLAN OF MERGER

IN WITNESS WHEREOF, the parties hereto do execute and deliver this Agreement as of the date first above written.

### COMPANY:

HIGHJUMP SOFTWARE, INC.

By /s/ Christopher D. Heim \_\_\_\_\_

Name: Christopher D. Heim Title: President

# PARENT:

3M COMPANY

By /s/ John F. Pohl

Name: John F. Pohl

Title: Division Vice President

# MERGER SUBSIDIARY:

STEELER MERGER CORPORATION

By /s/ Gregg M. Larson

\_\_\_\_\_

Name: Gregg M. Larson Title: Incorporator

### NEW LLC:

STEELER MERGER LLC

By /s/ Gregg M. Larson

\_\_\_\_\_

Name: Gregg M. Larson Title: Authorized Person

February 6, 2004

3M Company 3M Center St. Paul, MN 55144

Re: Registration Statement on Form S-3

#### Ladies and Gentlemen:

I have examined the Registration Statement on Form S-3 to be filed by you with the Securities and Exchange Commission on or about the date hereof (the "Registration Statement") in connection with the registration under the Securities Act of 1933, as amended, of 541,581 shares of common stock, par value \$0.01 per share (the "Shares") of 3M Company ("3M"). All of the Shares are issued and outstanding and may be offered for sale for the benefit of the selling stockholders named in the Registration Statement. I understand that the Shares are to be sold from time to time at prevailing prices or as otherwise described in the Registration Statement.

I have also examined the proceedings taken by you in connection with the issuance of the Shares.

It is my opinion that the Shares are validly issued, fully paid and  ${\tt non-assessable}$ .

I consent to the filing of this opinion as an exhibit to the Registration Statement and further consent to the use of my name wherever appearing in the Registration Statement, including the Prospectus constituting a part thereof, and any amendments thereto.

Very truly yours,

/s/ Gregg M. Larson
-----Gregg M. Larson
Assistant General Counsel

Securities and Exchange Commission 450 Fifth Street, N.W. Washington, D.C. 20549

### Commissioners:

We are aware that our reports dated April 21, 2003, July 21, 2003 and October 20, 2003, respectively, on our reviews of interim consolidated financial information of 3M Company and Subsidiaries (the "Company") for the three-month periods ended March 31, 2003 and 2002, the three- and six-month periods ended June 30, 2003 and 2002 and the three- and nine month periods ended September 30, 2003 and 2002, and included in the Company's Form 10-Q for the quarters ended March 31, 2003, June 30, 2003 and September 30, 2003, respectively, are incorporated by reference in the Company's Registration Statement on Form S-3, for the registration of 541,581 shares of the Company's common stock in connection with the amended agreement and plan of merger among the Company, Steeler Merger Corporation, Steeler Merger LLC, and HighJump Software, Inc.

/s/ PricewaterhouseCoopers LLP

Minneapolis, Minnesota February 6, 2004

### CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in this Registration Statement on Form S-3 of our report dated February 10, 2003, except as to Note 12, for which the date is May 22, 2003, relating to the consolidated financial statements, which appears in the 3M Company Current Report on Form 8-K dated May 23, 2003. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

/s/ PricewaterhouseCoopers LLP

Minneapolis, Minnesota February 6, 2004

#### POWER OF ATTORNEY

KNOW ALL MEN BY THESE PRESENTS, that each of the undersigned, being a director or officer of 3M Company (the "Company"), a Delaware corporation, hereby constitutes and appoints W. James McNerney, Jr., Patrick D. Campbell, Gregg M. Larson, Richard F. Ziegler, and J. Mark Borseth, and each of them, his or her true and lawful attorney-in-fact and agent, with full power of substitution and resubstitution, for him or her and in his or her name, place and stead in any and all capacities, to sign one or more Registration Statements under the Securities Act of 1933, as amended, on Form S-3 or such other form as such attorneys-in-fact, or any of them, may deem necessary or desirable, any amendments thereto, and all post-effective amendments and supplements to such registration statement, for the registration of the number of shares of the Company's common stock set forth in the registration statement, in such forms as they or any one of them may approve, and to file the same with all exhibits thereto and other documents in connection therewith with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done to the end that such Registration Statement or Registration Statements shall comply with the Securities Act of 1933, as amended, and the applicable Rules and Regulations adopted or issued pursuant thereto, as fully and to all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or any of them or their substitute or resubstitute, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned have subscribed these presents this 18th day of December, 2003.

/s	/	W	٠	J	am	es	1	1cl	Иe	rr	1e	у,	J1	r.							
_				_							-		 		 	 	_	 	 	_	

W. James McNerney, Jr., Chairman of the Board and Chief Executive Officer President and Chief Financial (Principal Executive Officer and Director)

- -----

/s/ Linda G. Alvarado

Linda G. Alvarado, Director

/s/ Edward A. Brennan - -----

Edward A. Brennan, Director

/s/ Vance D. Coffman

-----Vance D. Coffman, Director

/s/ Michael L. Eskew . .....

Michael L. Eskew, Director

/s/ Edward M. Liddy

. ..... Edward M. Liddy, Director

/s/ Robert S. Morrison

\_\_\_\_\_

Robert S. Morrison, Director

/s/ Patrick D. Campbell

Patrick D. Campbell, Senior Vice Officer (Principal Financial Officer)

-----

\_\_\_\_\_

/s/ Ronald G. Nelson

Ronald G. Nelson, Vice President

and Controller

/s/ Aulana L. Peters -----

Aulana L. Peters, Director

/s/ Rozanne L. Ridgway

Rozanne L. Ridgway, Director

/s/ Kevin W. Sharer

\_\_\_\_\_ Kevin W. Sharer, Director

/s/ Louis W. Sullivan \_\_\_\_\_

Louis W. Sullivan, Director